# Direction des Études et Synthèses Économiques

## G 2015 / 17

The redistributive effect of online piracy on the box office performance of American movies in foreign markets

Christophe Bellégo and Romain DE NIJS

Document de travail



Institut National de la Statistique et des Études Économiques

## INSTITUT NATIONAL DE LA STATISTIQUE ET DES ÉTUDES ÉCONOMIQUES

*Série des documents de travail de la Direction des Études et Synthèses Économiques* 

## G 2015 / 17

## The redistributive effect of online piracy on the box office performance of American movies in foreign markets

## Christophe BELLÉGO\* and Romain DE NIJS\*\*

NOVEMBRE 2015

Les auteurs remercient Michael L. ANDERSON, Luc BEHAGHEL, Fabrice BERNIER, Sofia BORGES, Benoît DANARD, Hadrien DE NIJS, Benjamin E. HERMALIN, Przemysław JEZIORSKI, Claire LELARGE, Laurent LINNEMER, Deborah MREJEN, John MORGAN, Minjun PARK, Corinne PROST, Juliette STEHLÉ, Felix VARDY, Mauricio VARELA, Michael VISSER, Patrick WAELBROECK pour leurs apports. Ce travail a également bénéficié des commentaires des participants aux séminaires du D2E à l'Insee, du Crest, de Paris School of Economics, de Toulouse School of Economics, du congrès annuel de l'EEA à Toulouse, de la conférence annuelle de l'EARIE à Milan, de la 7<sup>e</sup> journée de la culture et de la communication, et de la 8<sup>e</sup> conférence sur l'économie des TIC à Telecom Paris Tech.

\*\* École Polytechnique

Département des Études Économiques - Timbre G201 - 15, bd Gabriel Péri - BP 100 - 92244 MALAKOFF CEDEX - France - Tél. : 33 (1) 41 17 60 68 - Fax : 33 (1) 41 17 60 45 - CEDEX - E-mail : <u>d3e-dg@insee.fr</u> - Site Web Insee : http://www.insee.fr

Ces documents de travail ne reflètent pas la position de l'Insee et n'engagent que leurs auteurs. Working papers do not reflect the position of INSEE but only their author's views.

<sup>\*</sup> Insee, Crest. Département des Études Économiques - Division « Marchés et Entreprises » Timbre G230 - 15, bd Gabriel Péri - BP 100 - 92244 MALAKOFF CEDEX

## The redistributive effect of online piracy on the box office performance of American movies in foreign markets

#### Abstract

Using the French anti-piracy law known as Hadopi, we study the effects of online piracy on movie sales in theaters. Applying four estimation strategies at different levels of observations (country, town, consumer, and movie), we find that the introduction of the law is associated with a 9% increase in the market share of American movies but no expansion of the total market. We exclude supply side reactions by distributors as an explanation for this displacement effect that benefits American movies. The increase in the box office performance of American movies is primarily explained by the behavior of younger consumers.

**Keywords**: Anti-piracy law, Property rights, Online piracy, Business stealing, Demand expansion, Movie.

## L'effet redistributif de la loi Hadopi sur la part de marché des films américains en salle

### Résumé

Cette étude évalue la loi anti-piratage dite « Hadopi » pour étudier les effets du piratage sur les recettes des films en salle. À l'aide de quatre stratégies d'identification à différents niveaux d'observation (pays, villes, consommateurs, films), nous trouvons que l'introduction de la loi Hadopi est associée à une augmentation de la part de marché des films américains de 9 % mais sans augmentation de la demande totale pour les films en salle. L'analyse écarte les différentes réactions des distributeurs comme autres explications possibles à cet effet de substitution favorable aux films américains. La hausse des recettes des films américains américains est principalement expliquée par le comportement des jeunes consommateurs.

**Mots-clés** : Loi anti-piratage, droit de propriété, piratage sur Internet, redistribution des ventes, expansion de la demande, film.

Classification JEL : D12, K11, K42, L82

### 1 Introduction

Online piracy of cultural goods is a widespread phenomenon in developed countries and has important consequences for firms, consumers and public authorities (Oberholzer-Gee and Strumpf (2010), Waldfogel (2012a), Belleflamme and Peitz (2012)). According to Sandvine (2011), 19% (resp. 30%) of aggregate Internet traffic in North America (resp. Europe) was due to file sharing in 2011.<sup>1</sup> Society faces a major economic tradeoff between giving proper incentives to creators and allowing the public to benefit from the wider distribution of cultural goods with a marginal cost of nearly zero. Four countries (France, New Zealand, Sweden, and the United States) have recently enacted similar legislation to eliminate copyright infringement on the Internet.<sup>2</sup> The adoption and the implementation of these laws have been controversial and their intended effects highly discussed by policy makers.

This article intends to contribute new evidence to the debate by evaluating the effects of the French anti-piracy law (Hadopi) on the box office performance of movies. The Hadopi law was adopted at the end of 2009 and intended to reinforce Intellectual Property Rights for cultural goods.<sup>3</sup> This legislation is particularly intended to prevent Internet users from illegally downloading cultural goods on peer-to-peer file sharing services. Between September 2010 and December 2011, 824, 576 warning emails and 68, 341 registered letters were sent to users detected as infringers, which may have been a sufficiently substantial intervention to change individuals' behavior.

The market for movies in French theaters is well-suited to examining the effects of this anti-piracy law for several reasons. First, it is the first market to generate revenues for movies, and it conditions their success in subsequent markets (i.e., video and TV). Second, the market for movies in theaters is the main source of revenue for producers and distributors in France.<sup>4</sup> Third, foreign earnings represent a substantial share of the total gross of the U.S. motion picture industry (62.5% of total lifetime grosses of the top-10 most successful U.S. movies in 2013), and France is often among the yearly top 3 largest markets for U.S. exports.<sup>5</sup> Fourth, prices are fixed in this industry, i.e., there is no movie-by-movie pricing (Einav and

<sup>&</sup>lt;sup>1</sup>Sandvine is a networking equipment company. Aggregate traffic is an average of downstream traffic and upstream traffic weighted by their sizes.

 $<sup>^{2}</sup>$ The Copyright Alert System (CAS) has been implemented in the United States in February 2013. It is a graduated response system including six strikes that is very similar to the French Hadopi law. Unlike France, New Zealand, or Sweden, the CAS does not constitute a law but is the result of an agreement between right holders and Internet service providers.

<sup>&</sup>lt;sup>3</sup>Hadopi is a "law promoting the distribution and protection of creative works on the Internet" see http://www.legifrance.gouv.fr: "LOI n° 2009-669 du 12 juin 2009 favorisant la diffusion et la protection de la création sur Internet"

 $<sup>^{4}</sup>$ It represents 1,370 million euros and 207 million admissions in 2010 in France. DVD retail accounts for revenues equal to 1,200 million euros and 87 million transactions in 2010.

<sup>&</sup>lt;sup>5</sup>The top-10 U.S. movies in the U.S. market in 2013 were The Hunger Games: Catching Fire, Iron Man 3,

Orbach (2007)). This prevents any finely tailored pricing reaction to the law, and we can measure sales by the number of tickets sold and not in monetary amounts. Fifth, data on both the demand (number of tickets sold) and the supply side (number of screens) of the industry are available, thus enabling us to distinguish supply-side from demand-side reactions to the law. Sixth, data on local geographic and consumer-level sales of movies are accessible, which allow us to perform a comprehensive analysis. Seventh, some evidence show that the supply of illegal copies only exists for U.S. films, but not for French ones, during their theatrical exhibition in France. This entails potential asymmetrical effects of online piracy that one could not study in other subsequent markets such as the video market, where both types of films are illegally available on the Internet.

We adopt four empirical strategies at different levels of observation to estimate two potential effects of Hadopi: (1) a demand expansion effect, using the total number of admissions as the outcome of interest, and (2) a redistributive effect among movies, using the market share of American movies as the outcome of interest. The four approaches are mutually reinforcing and complementary. Overall, these estimations indicate that the law is associated with an increase in the market share of American movies at the expense of other (domestic) movies without expanding the total market. Furthermore, we propose a simple model to derive the predictions that we use to understand the consumer behavior underlying the piracy effect and to approximate its impact on welfare in the short-term.

First, we conduct an investigation at the town level within France, using original data on the diffusion rate of broadband Internet, and movie sales. We find that towns with a higher level of online piracy (approximated by a higher penetration rate of broadband Internet) experienced a higher increase in American movie admissions after the introduction of the Hadopi law. There is no evidence of a significant market expansion effect associated with the law. These findings indicate that the Hadopi law had a pure redistributive effect between French and American movies. By examining the same movies across different towns, this approach excludes an important competing explanation: that American (resp. French) films could have experienced a positive (resp. negative) shock in France concomitant with but unrelated to the Hadopi law.

Second, we compare American movie sales to French movie sales within France. This approach shows that U.S. major film studios did not react to the Hadopi law by changing the quantity or the quality of the movies they export to France. Therefore, the effect of the law is very unlikely to be mainly driven by

Frozen, Despicable Me 2, Man of Steel, Gravity, Monsters University, The Hobbit: The Desolation of Smaug, Fast & Furious 6, Oz The Great and Powerful. Actually, this figure understates the true share that foreign markets represent, as some of those movies were yet to be released in some major foreign markets in 2013 (Source: www.boxofficemojo.com). The other largest markets for American films are Canada, Germany, Italy, Japan, and the United Kingdom/Ireland (Source: Motion Picture Association of America (MPAA)).

a supply reaction by the industry. This approach also confirms that the law is associated with an increase in the market share of American movies.

Third, we compare the performance of American movies across countries, using several relevant European countries as a control group for France. We find no supply reaction by American distributors in France, in terms of either the speed or quantity of exported films, compared with other countries. In addition, we again confirm that the sales of American movies substantially increased after the implementation of the Hadopi law.

Finally, using consumer-level data, we study movie sales from different groups of consumers. We find that sales to younger consumers, who are more likely to illegally download movies, have increased for American movies since the Hadopi law was passed, compared with French movies and older consumers.

The positive effect of the Hadopi law on American movies is consistent with the fact that American movies are much more subject to online piracy than are other movies when they are released in foreign theaters. American movies are often released in the United States earlier than in foreign markets (including France), and criminal networks that supply the online market with illegal copies are more organized and active on the American continent than in European countries, thereby allowing greater proliferation of their illegal copies on the Internet. During their theatrical exhibition in France, there is (almost) no illegal version of French films, but there are many illegal versions of American films. Consequently, when an anti-piracy law is enacted and makes it riskier to download films on the Internet, sales of American movies in theatres could increase.

The absence of a market expansion effect can be rationalized by consumers' tight budgets and time constraints that prevent any meaningful expansion of the market for movies in theaters. This reflects the idea of Heath and Soll (1996), recently refreshed by Hasting and Shapiro (2013), that individuals set budgets for categories of expenses and then track expenses relative to their budget, in an attempt to avoid diverging from their usual level of expenditures in a given categorie. Attending the movies is more time-consuming than watching pirated copies at home, because, for instance, of travel costs, waiting time, impossibility to stop watching a movie at any time and come back to it later on, and higher difficulty to stop watching a movie that one does not enjoy. Moreover, attending the movies is primarily a social practice that consumers would not reduce. According to the Centre National du Cinéma et de l'image animée (CNC), approximately 9% of individuals attended the movies alone during the period 2008 - 2011.

Specifically, the underlying behavior would be as follows. Without an anti-piracy law, some people illegally consume American movies online and legally watch domestic (French) movies in theaters because illegal copies of American movies are easily available on the Internet during their theatrical exhibition, but

this is not the case for other (domestic) movies. When an anti-piracy law is implemented, they shift to legal consumption of American movies in theaters because it is now too risky to illegally obtain them online, and because they tend to prefer American films. Furthermore, as a side effect, they reduce their attendance for non-U.S. movies in theaters, due to budget and time constraints. As a result, the total industry profit remains unchanged but is redistributed: some sales of domestic (French) movies are displaced to American movies. However, consumers are not compensated for the loss of domestic films that they no longer watch. According to back-of-the-envelop calculation, this loss translates into a decrease in consumer welfare of approximately 73 to 94 million euros per year in the short-term.

The remainder of this article is organized as follows. Section 2 reviews the relevant literature. Section 3 provides some background on the online piracy of movies. Section 4 describes the data. Section 5 presents the results of the four complementary empirical approaches. Section 6 provides straightforward welfare implications and concludes.

### 2 Relevant literature

This article is related to the empirical literature on the effects of piracy on legal sales. This literature has been expanding rapidly, and only the most relevant articles to the current work are discussed in this section. Readers are referred to Waldfogel (2012a), Smith and Telang (2012), and Waldfogel (2012b) for more complete reviews and discussions of the literature. These findings also contribute to the general empirical literature on how the Internet affects the offline activity of individuals (Gentzkow (2007), Liebowitz and Zentner (2012), and Zentner et al. (2012)).

Three articles are closely connected to ours. Using a difference-in-differences approach across different countries with weekly aggregate iTunes sales, Danaher et al. (2014) find a positive effect of Hadopi on digital music sales in France. They also find evidence that this positive effect began before the law was passed, when Hadopi was much discussed in the media because of saliency. Adermon and Liang (2014) evaluate the effects of an anti-piracy law in Sweden, similar to the French Hadopi law, on music and movie sales. Applying difference-in-differences comparisons with Finland and Norway as control groups, they find that the Swedish law substantially decreased Internet traffic and increased music sales during the subsequent six months but had no significant effects on movie sales in theaters or on video. Finally, Bhattacharjee et al. (2006) study the effect of legal threats from the Recording Industry Association of America (RIAA) toward Internet users in 2003. Adopting an event study, they suggest that individuals have substantially decreased the number of files shared, although a wide variety of files remained available on the Internet after these actions.

Our article extends these three papers' contributions. First, we estimate the redistributive effects of an anti-piracy law by differentiating between different types of movies according to the difference in the level of online piracy, using the nationaly as a proxy. Then, comparing movie sales in towns with different levels of online piracy, we show that our results are not driven by shocks to the market for movies that are concomitant with but unrelated to the Hadopi law and, hence, that the estimated causal effect is much more convincing. Next, we cautiously investigate potential short-run supply-side reactions to the law in our international comparison analysis.<sup>6</sup> In addition, we devote careful attention to the choice of control group using the method developed by Hilger et al. (2011).

Our international approach also goes a step further and is conducted at the movie level. This enables us to control for movie characteristics and for competition, which are important determinants of box office performance. Furthermore, we deepen our understanding of the Hadopi effect using consumer-level data

<sup>&</sup>lt;sup>6</sup>See Alcala and Gonzalez-Maestre (2010) for a similar theoretical argument.

to identify the type of consumers embedded in this effect. Finally, we estimate the welfare variation using a simple model and back-of-the-envelope calculations.

We find that anti-piracy laws are not associated with legal demand increase in theaters. This result is similar to that of Zentner (2010), who finds no statistically significant link between high-speed Internet penetration (a proxy for online piracy) and total box office revenues in a set of countries. It is also consistent with Bai and Waldfogel (2012) who find, using surveys on students, a small negative effect of online movie piracy on legal consumption in theaters. Regarding the recorded music market, Oberholzer-Gee and Strumpf (2007) also obtain an estimated effect of file sharing on sales that is not statistically distinguishable from zero. Peukert et al. (2013) and Danaher and Smith (2013) study the effects of the shutdown of Megaupload on movies' box office performance and digital sales, respectively. Danaher and Smith (2013) find a small increase in digital revenues (7%) but no effect on physical revenues. Peukert et al. (2013) show that only large blockbusters have taken advantage of the closing of Megaupload in theaters, which is consistent with the redistributive effect we find in this paper.

Our approach is also related to that of Liebowitz (2008), who compares the sales of music albums in cities with different levels of broadband Internet coverage rate as a proxy for file sharing. He finds that online piracy has a large negative impact on music sales over the period 1998-2003. In comparison, we additionally use an anti-piracy law as an exogenous shock to online piracy, consider several other empirical approaches to exclude other possible explanations for our result, and examine the motion picture industry during a recent period. In the context of the movie industry, several other empirical strategies find a displacement effect between illegal sharing and legal purchases, see De Vany and Walls (2007), Hennig-Thurau et al. (2007), Rob and Waldfogel (2007), Danaher et al. (2010), and Danaher and Waldfogel (2012). For papers on the recorded music industry, see Liebowitz (2006), Rob and Waldfogel (2006), Zentner (2006), and Bhattacharjee et al. (2007). They also demonstrate a negative effect of online piracy on album sales.

This paper is also linked to studies of the theoretical effect of file sharing on sales and industry profit. Overall, this literature shows that this effect is ambiguous; users could substitute illegal copies for legal consumption, which would increase the marginal cost of producing original units. However, file sharing could also increase profit under certain conditions (Conner and Rumelt (1991), Takeyama (1994), Takeyama (1997), Bakos et al. (1999), Varian (2000), Gopal et al. (2006), Peitz and Waelbroeck (2006), Rob and Waldfogel (2006), Jain (2008)). See for other recent papers in this literature Harbauch and Khemka (2010), Vernik et al. (2011), and Galbreth et al. (2012).

### 3 Description of online piracy of movies and of Hadopi

### 3.1 The origins of illegal copy of movies

The two major sources of illegal copies are camcording and video copies. According to the MPAA, camcording accounts for the highest share of newly released movies that are pirated. Camcording refers to thieves who use a digital recording device in a movie theater.<sup>7</sup> Illegal versions can also come from the video release of a movie through a DVD rip or Blu-ray rip. Approximately two or three weeks before a film's video release, DVD and Blu-ray copies are sent to distributors. Thereafter, it is very difficult to identify the origin of the leak, and video copies become easily available on illegal markets.

There are other sources of illegal copies that are less frequently used. Illegal copies of films are occasionally made from advance copies used for screening and marketing purposes called 'screeners'. Most of these can be traced through watermarking, which prevents most art theft, but pirates are able to circumvent this protection measure. Illegal copies can also be made from a stolen film print or digital file from a theater, film depot, courier service, or other industry-related facility for making or financing films.

Camcording and the theft of advance copies are very common practices in North America but not in France.<sup>8</sup> In addition, American movies are often released sooner in the U.S. than in France. In 2008, the median (mean) difference between the French theatrical release and the American one was 47 (59) days. Though is has decreased over time, this difference has remained largely positive. It was 47 (91) days in 2009, 40 (87) days in 2010, and 33 (57) days in 2011.

As a consequence, an illegal copy of an American movie is often available before its theatrical release in France, but this is not the case for most French movies. Thus, American movies in France are more exposed to piracy because of the supply of illegal copies on the Internet. Some casual empirical figures support these arguments. In October 2008, before the Hadopi law, the ALPA released a report on illegal downloading of movies in France.<sup>9</sup> Tables 1 and 2 draw on this report. They clearly illustrate that U.S. movies tend to be much more frequently illegally downloaded than French ones and are downloaded sooner (and in most cases before their release in France).

<sup>&</sup>lt;sup>7</sup>Studio and theater owners have increased security and surveillance in theaters worldwide, and technologies such as Coded Anti-Piracy films or watermarking films are used to identify the source of illegal copies to prevent camcording.

<sup>&</sup>lt;sup>8</sup>According to the ALPA, the main French association combating audiovisual piracy, there are almost no illegal copies coming from the theatrical exhibition of movies, and very few pirate teams specialized in camcording in France between 2007 and 2011. Other interviews with experts from the MPAA and from the Hadopi's government agency also confirm this, although it is difficult to obtain definitive figures.

<sup>&</sup>lt;sup>9</sup>Le téléchargement illégal des oeuvres cinématographiques francophones, ALPA, 2008.

The difference in online piracy between U.S. and French films does not appear to come from the demand side. To illustrate this point, the total numbers of admissions in French theaters for the top-3 downloaded French movies are 20 400 000, 1 500 000, and 5 300 000 (from Bienvenue chez les Ch'tis, Persepolis, and La Môme, respectively), whereas for the top-3 downloaded U.S. movies, they are 1 200 000, 300 000, and 2 600 000 (from Jumper, Cleaner, and Iron Man, respectively). These figures show that the downloading of U.S. movies is relatively high while their demand in theaters (as measured by total admissions) is comparable to or lower than that for French movies.

		\ \	,
title	number of daily downloadings	date of French release	date of first detected illegal downloading
Bienvenue chez les Ch'tis	9,800	27/02/2008	05/03/2008
Persepolis	3,500	27/06/2007	13/10/2007
La Môme	1,400	14/02/2007	15/10/2007
Disco	4,400	02/04/2008	10/04/2008
Survivre avec les loups	2,400	16/01/2008	25/01/2008
JCVD	2,400	04/06/2008	06/06/2008
15 an et demi	800	30/04/2008	08/05/2008
Chasseurs de Dragon	600	26/03/2008	07/04/2008
Sans arme, ni haine, ni violence	500	16/04/2008	28/04/2008
L'Homme sans âge	500	14/11/2007	26/11/2007

Table 1: Top-10 downloaded French movies (June 2008)

Source: ALPA statistics on counterfeit movies downloading on peer-to-peer networks, powered by Advertiso and THOMSON Image & beyond.

title	number of daily downloadings	date of French release	date of first detected illegal downloading
Jumper	24,500	20/02/2008	18/02/2008
Cleaner	14,300	14/05/2008	22/05/2008
Iron Man	14,000	30/04/2008	05/05/2008
Cloverfield	13,800	06/02/2008	25/01/2008
Horton	13,000	02/04/2008	17/03/2008
Transformers	12,900	25/07/2007	13/10/2007
Juno	12,500	06/02/2008	20/12/2007
There will be blood	12,400	27/02/2008	11/01/2008
Bee movie	11,900	12/12/2007	25/11/2007
No country for old men	11,900	23/01/2008	28/11/2007

Table 2: Top-10 downloaded U.S. movies (June 2008)

Source: ALPA statistics on counterfeit movies downloading on peer-to-peer networks, powered by Advertiso and THOMSON Image & beyond.

### 3.2 The Hadopi law

The French Hadopi law is a "law promoting the distribution and protection of creative works on the Internet". The law was presented to the National Assembly in March 2009, where it was at first supported and then rejected. The law was enacted in October 2009 after more than one year of debate at the Senate, the National Assembly and the Constitutional Council (Danaher et al. (2014)). Finally, the first emails

are sent by Hadopi in September 2010.<sup>10</sup> Hadopi is also the acronym for the government agency created to administer this law.

Hadopi was created for three official purposes: implementing a graduated response against online pirates, creating awareness/education among consumers regarding online piracy issues, and fostering legal online supply. Copyright holders, beneficiary owners or their representatives such as ALPA detect an infringement using peer-to-peer file sharing and inform the Hadopi authority, reporting the IP address of the infringer.<sup>11</sup> The Hadopi agency checks this report and sends the IP address to Internet service providers to identify the Internet access subscriber. Then, the Hadopi agency may initiate a graduated response in a 'three-strikes' procedure:

- 1st strike: an email message is sent to the incriminated Internet access subscriber as a warning (824,576 emails were sent between September 2010 and December 2011, and more than 4 million emails until 2015).
- 2nd strike: if, in the six months following the first strike, a repeat infringement is detected, a registered letter is sent to the infringer as a second warning, with similar content to the first email message (68,341 registered letters were sent over the same period, and more than 400,000 registered letters until 2015).
- 3rd strike: if, in the one year following the second strike, a repeat infringement is detected, a second registered letter is sent to the infringer, informing him or her that the facts of the case are now subject to legal proceedings. The infringer can be subjected to penalties such as a fine and a loss of Internet access for one month to one year, in which case the infringer is blacklisted from Internet service providers.<sup>12</sup>

The Hadopi law directly targets illegal peer-to-peer downloads. It may have deterred Internet users from downloading after receiving a warning from the Hadopi agency. The law may also have had an awareness and an educational effect on people who download illegally, who may in turn shift to legal con-

 $<sup>^{10}</sup>$ Decree n° 2010-695 was published 26 July 2010; it was the last piece of legislation needed to implement the graduated response.

<sup>&</sup>lt;sup>11</sup>The Hadopi agency is only empowered to initiate proceedings against illegal peer-to-peer file sharing. Technically, the Hadopi agency only tracks the 100 most popular and most downloaded movies in any given period. This list is not public and is regularly renewed.

 $<sup>^{12}</sup>$ Since the law was approved in 2009, only one Internet user has been suspended (for 15 days) and was also fined EUR 600. The fine cannot exceed EUR 1500. However, the Hadopi agency has obtained 114 legal judgements within the framework of the graduated response. These figures appear low, but the graduated response is not designed to punish people and is mainly intended to educate people that they should not download illegally. Cutting off Internet access has been removed from the law in July 2013.

sumption. Conversely, infringers may have simply switched to streaming technologies or direct downloads, which are not monitored by Hadopi, but we provide evidence below showing that this was not the case.

### 3.3 A business stealing effect is plausible

The law is likely to induce a business stealing effect for several reasons. First, many sources tend to show that there was a decline in peer-to-peer use from September 2010 to December 2011, when the graduated response started being implemented in France. Nielsen noted an important reduction of 50% in the audience levels of websites offering links (see graph 1(a)), whereas Peer Media also measured a drop of 43% in the illegal sharing of films on peer-to-peer networks in France during 2011 (see graph 1(b)).<sup>13</sup> This decline is consistent with the considerable number of messages sent by the Hadopi agency. Moreover, in a survey conducted in 2011 (Hadopi (2011)), 72% of warned Internet users reported that they had reduced or completely stopped their illegal usage following this warning. Finally, Mediametrie/NetRatings report no substantial increase in streaming and direct download services (see graph 4 in Appendix B), and Darmon et al. (2014) show using a survey of 2000 individuals conducted in 2012 that 78% of Internet users rightly believed that the Hadopi agency monitored illegal peer-to-peer downloads, but 68% (respectively 37% and 12%) of them also thought that the Hadopi monitored illegal direct downloads (respectively illegal streaming and illegal offline file sharing) while this is not the case. This body of evidence converges to show a decrease in online piracy that has mainly come from peer-to-peer channels and without important transfers to other piracy channels.

Data from the French national accounts on household consumption show that the share of consumption of movies in theaters in the shopping basket of the average French household has been stable since 1990 (see graph 6 in Appendix B). This supports the view that budget and time constraints might constitute important limitations to the expansion of the market for theatrical exhibition. Next, a survey conducted by *Observatoire de la satisfaction* shows that the distribution of audiences for American movies and French movies overlap significantly in terms of gender and age (see graphs 1(a), and 1(b) in Appendix B).<sup>14</sup> As a consequence, a large number of consumers are likely to enjoy watching both French and American movies. Finally, several pieces of evidence show that many French consumers tend to prefer American movies to French ones, which is a necessary condition to obtain a business stealing effect (the production budget, marketing expenditures, number of screens during the release week, and consumer ratings are significantly higher for American movies than for French ones; see Appendix B for more precise figures).

<sup>&</sup>lt;sup>13</sup>Peer Media is an American company providing anti-piracy services targeted at peer-to-peer networks.

<sup>&</sup>lt;sup>14</sup>Observatoire de la satisfaction is an independent research and polling institute specialized in the movie industry.

#### Figure 1: A clear decline in illegal downloading in France

(Sources: (a) IFPI / Nielsen, "Digital Music Report 2012", January 2012, total duplicated audience offering links to P2P files and applications, across approximately 40 P2P services, in thousand of unique visitors. (b) Peer Media Technologies, February 2012, sharing of internationally-observed films, change between December 2009 and December 2011, in millions of downloads initiated. Sample of 200 to 300 recent films in rotation.)



to peer-to-peer files and applications

(a) The fall in audience levels of websites offering links (b) The drop in illegal sharing of movies on peer-to-peer networks in France compared to the rest of the world

### 4 Data sources and descriptive statistics

In this paper, we conduct four empirical analyses by gathering many unique datasets that we describe in this section in the order in which the results are presented.

### 4.1 The town-level data

To implement a town-level approach, we use two sources of data, which we match through the Insee geographical codes for each town. These two sources are described below.

• Town-level Internet infrastructure and use data

The Internet data contain information on the infrastructure of high-speed Internet connections in France during the fourth quarter of 2011. They also contain the quarterly records of the number of households that have subscribed to high-speed Internet at every exchange point over the period 2008-2011. Using this dataset, we define the broadband coverage rate as the fraction of connections with download speeds higher than 512 kbit/s over the total number of connections available in a town and the broadband Internet use rate as the fraction of households that have subscribed to connections exceeding 512 kbit/s over the total number of connections for a given town. All details concerning the method to compute these indicators are available in Appendix C. Throughout this paper, high speed Internet is defined as Internet connections with download speeds that exceed 512 kbit/s. The Internet data come from Arcep, the French agency responsible for regulating telecommunications in France.

Table 3 presents the penetration of broadband in terms of coverage rate and use rate. By the end of 2008, the 68 towns in our dataset were well equipped with high-speed Internet infrastructure: 95% of these towns are supplied with at least 89.6% high-speed connections. This infrastructure equipment did not change during the period 2008-2011. The variation in the high-speed Internet use rate is more important than the variation in the coverage rate. The observation period also highlights the adoption of high-speed Internet by the households. The 5th percentile of the use rate was 46.7% in 2008, while it was 56.5% in 2011.

High speed Internet		Perce	$\mathbf{n}$ tiles	
coverage and use	p05	p25	p75	p95
Coverage rate (%)	89.6	90.5	92.3	93.3
Use rate in 2008 $(\%)$	46.7	49.6	53.9	59.2
Use rate in 2009 $(\%)$	49.0	52.5	57.1	62.1
Use rate in 2010 $(\%)$	53.5	56.5	60.6	64.6
Use rate in 2011 (%)	56.5	58.8	62.4	66.1

Table 3: Descriptive statistics for the local analysis: penetration of Internet

• Town-level movie sales data

The second dataset contains the box office performance of movies in the same 68 middle-size towns in France by movie theater, week, and movie, between February 2009 and August 2011.<sup>15</sup> The town level movie sales data come from *Médiamétrie*, a French audience measurement company.

Table 4 displays summary statistics about movie sales over the 68 towns.

	· · ·					
	Before	Hadopi	After Hadopi			
	Towns with	Towns with	Towns with	Towns with		
	Internet coverage rate	Internet coverage rate	Internet coverage rate	Internet coverage rate		
	under the median	above the median	under the median	above the median		
U.S. movies admissions	19,952.3	23,437.1	25,124.3	32,949.3		
	(18,007.6)	(25, 286.9)	(19,047.2)	(28, 627.0)		
French movies admissions	13,730.3	16,945.5	14,841.6	19,874.0		
	(11, 235.4)	(16, 515.8)	(12, 193.8)	(18, 249.9)		
Standard errors in parentheses						

Table 4: Descriptive statistics for the local analysis: average movie sales

### 4.2 The French movie sales data

This data set consists of all movies released on at least 100 screens in France between January 2006 and December 2011. The data include the number of admissions and the number of screens during the release week in theaters in France. The data also contain rich information on movie characteristics: genre, nationality, producers, distributors, average press reviews, and consumer reviews on Allocine.fr, age restriction, art house movie classification, and awards in two main film competitions (Cannes Festival and Oscars). Finally, the data include total marketing expenditures. Most of the data are provided by the CNC, the French public administrative organization under the authority of the Ministry of Culture and Communication, which is responsible for monitoring the French movie industry, and from websites. Marketing data come from TNS Sofres, which is a leading market research and market information group.

Table 5 provides the main descriptive statistics for French and American movies. French movies are released on fewer screens than are American ones. They are also less successful than American movies: the admissions per screen are 810 for French movies and 925 for American movies. This feature is consistent with higher consumer ratings for American movies (with an average of 3.15 on a scale of 5) than for French

<sup>&</sup>lt;sup>15</sup>Agen, Albi, Amiens, Angers, Angoulème, Arras, Belfort, Blois, Boulogne sur Mer, Bourges, Brest, Brive la Gaillarde, Caen, Castres, Chalon sur Saone, Chalons en Champagne, Chambery, Charleville Mézières, Chartres, Cherbourg, Clermont Ferrand, Colmar, Créteil, Dijon, Dunkerque, Elbeuf, Epinal, Evreux, Haguenau, La Rochelle, Laval, Le Havre, Le Mans, Lorient, Maubeuge, Meaux, Metz, Montargis, Montauban, Montpellier, Mulhouse, Nancy, Nantes, Narbonne, Nevers, Nimes, Niort, Orléans, Pau, Périgueux, Poitiers, Reims, Rennes, Romans sur Isère, Rouen, Saint Brieuc, Saint Etienne, Saint Nazaire, Saint Omer, Salon de Provence, Strasbourg, Tarbes, Thionville, Tours, Troyes, Valenciennes, Vannes, Villefranche sur Saone.

movies (with an average of 2.87). Finally, 32% of French films are classified as art-house whereas only 13% of American films are classified as art-house.

. .

- 11

Table 5: Descriptive statistics						
	U.S.	movies	Frencl	n movies		
	Mean	Std. Dev.	Mean	Std. Dev.		
Number of obs.	575		456			
1st week admissions/screens	927	582	814	542		
1st week admissions	386,582	452,890	303,546	415,199		
1st week screens	363	195	326	162		
Total admissions	984,314	1,310,327	898,626	$1,\!667,\!620$		
Art and house	0.13	0.34	0.32	0.47		
Average user rating	3.15	.48	2.87	0.58		
Average Press rating	2.82	0.80	2.92	0.74		

Figure 2 plots the ratio of admissions in the release week on the corresponding number of opening screens for French and American movies. Data are averaged by quarter. The media and national awareness of Hadopi began in the second quarter of 2009 (Danaher et al. (2014)), before the law was adopted. After this period, the ratio of American movies substantially increases and remains larger than the French counterpart in all subsequent quarters. This increase was sustained after the actual adoption of the law and the beginning of the enforcement of the graduated response. This graphical evidence suggest a positive association between Hadopi and U.S. movies' box office performance.

Figure 2: The increase in the performance of U.S. movies after the Hadopi law



(a) Admissions per screen between 2006 and (b) Admissions per screen between 2006 and 2011 (averaged by quarter) 2011 (averaged by semester)

#### 4.3 The international movie sales data

The data consist of weekly box office gross in 18 European countries during the period beginning in January 2007 and ending in December 2011. The data only include movie-level box office revenues for the top-10 or top-20 movies (depending on the country, but the criteria do not change over time for a given country) for a defined period (3-, 4-, or 5-day "weekends"). The data set contains the name of the movie, its distributor(s) in different countries, the release date in each country, and the number of opening screens and box office gross in U.S. dollars. The data were collected in September 2012 from ScreenDaily.com. In addition, we have annual information by country on GDP per capita and broadband penetration. Upon close inspection, the data are reliable and are provided in a common format across our period of interest for nine countries: Belgium, Finland, France, Germany, Italy, the Netherlands, Spain, Switzerland, and the United Kingdom. Movie revenues are converted into the number of tickets sold based on the current conversion rates and average ticket prices. Data cover 85% to 90% of the total number of admissions.

### 4.4 Consumer-level data

In France, a consumer has the option to subscribe to unlimited access movie cards issued by movie theater companies. This annual contract costs approximately 20 euros per month. The monthly fee is automatically debited to a bank account until cancellation. This card provides unlimited access to the movies exhibited in the theaters that accept the card. It includes all theaters that are part of the chain issuing the card and the independent theaters that have an agreement with that chain. The other option for a consumer to watch a movie in a theater is to pay approximately 10 euros per movie or to purchase a five-ticket pass for 40 euros. Note that students, unemployed persons, and seniors also have access to reduced-fare tickets.

We gathered unique sales data from three theaters in two medium-sized French cities that belong to the same chain and that all accepted the unlimited access card between 2008 and 2011.<sup>16</sup> These data contain the identity of the cardholder, the movie, the theater, and the date of each transaction and visit.

The combined market share of the three theaters accounts for between 80% and 90% of the market of the respective city. Another independent movie theater specialized in art-house movies is present in each city. We do not have sales data for those independent art-house movie theaters, but because the three movie theaters we use clearly dominate the local markets, we are confident that no competitive effect interferes with the subsequent analysis.

The unlimited access cards were introduced in 2003 and in 2006 in each city. The economic success of

<sup>&</sup>lt;sup>16</sup>In the first town, there are two large theaters located in the inner city and in the suburbs. In the second town, there is one theater located in the inner city.

this program has been substantial. Graphs 3(a) and 3(b) display the monthly growth rate and the monthly number of consumers that have used their card at least once in the three theaters. No sharp change in the number of consumers with cards appears at the time that the Hadopi law was passed. Therefore, we assume that there was no sudden reaction by consumers who would have subscribed to the unlimited access card in response to the law. However, the choice of subscribing to this service remains endogenous. As the number of cardholders has increased consistently over time, the graphs also show that the number of cardholders has increased more quickly for younger consumers than for older consumers.

Figure 3: The number of consumers holding an unlimited access card is growing over time, in three theaters of two medium-size cities of France



(a) Monthly growth rate of the number of (b) Monthly number of unique members unique members consuming movies consuming movies

Table 6 presents the percentage of U.S. films watched by consumers with card. This percentage is higher after the law was adopted. It is particularly the case for younger consumers, aged between 10 and 29, who generally download more than older consumers.

Table 6: Percentage of U.S. movies watched by consumers with unlimited access card

	Before Hadopi	After Hadopi
All consumers with card	51,2%	53,8%
Consumers aged between 10 and 29	58,2%	60,9%
Consumers aged between 30 and 49	52,4%	$53,\!4\%$
Consumers aged between 50 and 80	37,9%	39,4%

### 5 Empirical strategies and results

In this section, we develop four empirical analyses to assess the effects of the anti-piracy law from different perspectives. We summarize the main results of each analysis in the form of their pros and cons in Table 7.

Throughout the study, we test whether the law had an effect using an indicator that equals one beginning in November 2009 (hereafter *Hadopi*), when the law was adopted, and zero otherwise. In addition, Danaher et al. (2014) argue that the Hadopi law began to have an effect on digital music sales in March 2009, when the law had yet to be enacted but was highly visible in the public debate. As a consequence, to test whether such an effect is also present for movie sales, we allow for heterogeneous effects of the Hadopi law in our estimations with respect to three periods of time:

- A first indicator for the period between April 2009 and October 2009 (hereafter *Hadopi*1), which corresponds to the period between the beginning of the debate among lawmakers and the adoption of the first version of the law.
- A second indicator for the period between November 2009 and August 2010 (hereafter *Hadopi2*), which is the period when the law was first adopted.
- A third indicator for the period between September 2010 and December 2011 (hereafter *Hadopi*3), which is the period during which the graduated response came into force.

#### 5.1 Analysis 1: Town-level approach

#### 5.1.1 Empirical strategy

To examine the effects of the Hadopi law on the market for movie theaters in France, we compare box office performance in towns with a high-level of piracy to that in towns with a low-level of online piracy. This empirical strategy allows us to purge the estimate of the law's effect from common shocks at the national level in France that would be concomitant with Hadopi. For instance, these coincident shocks could have been the release of unpredictably successful American movies or the release of poorly performing French movies.

We operationalize the level of online piracy before Hadopi in each town using broadband Internet use as a proxy, as in Liebowitz (2008) and Bhuller et al. (2013). Our empirical approach assumes that towns with high broadband Internet use experience more illegal downloads than towns with low broadband Internet use. As a result, towns with a higher level of Internet use should have been more affected by the Hadopi

	Main effects identified	Advantages	Limits	Results
(1) Local comparisons: towns with high level vs. towns with low level of online piracy	Local market expansion effect and business stealing effect in towns with high level of online piracy compared to ones with low level of online piracy	- Main effects of the Hadopi law are captured Rules out simultaneous unrelated shocks specific to U.S. or to French movies	No conclusion at the national level: localized effect on middle-size towns	- No market expansion ef- fect but business stealing effect - The market share of U.S. movies increases by 0.8% with a one per- centage point increase of the broadband internet use rate
(2) National comparisons: U.S. movies vs. French movies in France	Relative business stealing effect of U.S. movies toward French movies	<ul> <li>Checks for supply re- actions from U.S. film distributors compared to French film distributors</li> <li>Controls for simultane- ous unrelated shock symet- ric on all movies in France</li> </ul>	<ul> <li>No conclusion on market expansion effect and pure business stealing effect</li> <li>Does not check for simul- taneous unrelated shocks specific to U.S. or to French movies</li> </ul>	<ul> <li>Relative increase of U.S. films admissions by 10% relatively to French films</li> <li>No reaction from the supply side of U.S. movies compared to French movies, except a decrease in the production budget, a decrease in user ratings, but an increase in the marketing expenditures</li> </ul>
(3) International comparisons: France vs. other european countries	Market expansion effect in France and business stealing effect from U.S. movies in France, compared to other countries	<ul> <li>Main effects of the Hadopi law are captured</li> <li>Checks for supply side re- actions of film distributors of U.S. movies in France compared to other euro- pean countries</li> </ul>	- Does not control for simultaneous unrelated shocks in France in gen- eral, or specific to U.S. or French movies - Does not check for sup- ply side reactions of U.S. film distributors compared to French film distributors, in France	<ul> <li>No market expansion effect but business stealing effect: market share of U.S. movies increases by 8%</li> <li>No change in the speed nor in the quantities of exports of U.S. films to France</li> </ul>
(4) Consumer comparisons: consumers with high propensity vs. those with low propensity to download	Relative business stealing effect of U.S. movies toward French movies, between consumers belonging to different age groups	<ul> <li>Distinguishes what type of consumers are influ- enced by the new regula- tion</li> <li>Provides an insight on the mechanism at work</li> </ul>	<ul> <li>No conclusion on total market expansion effect</li> <li>Very localized data col- lection</li> </ul>	<ul> <li>Relative increase of U.S. films admissions compared to French films admissions</li> <li>Increase by 20% for con- sumers aged 10-29 and by 10% for consumers aged 30-49, compared to con- sumers aged 50-80</li> </ul>

### Table 7: Recap of the advantages and limits of each empirical strategy

law. Indeed, high-speed Internet facilitates and reduces time for downloading large files such as movies. A 1 Gb file takes 17 hours to download with a 128 kbit/s connection but only 4 hours with a 512 kbit/s connection and 1 hour with a 2000 kbit/s connection.

Weekly movie sales in a theater can be very erratic and depend on the presence of blockbusters. These volatile variations in sales are not linked to the Hadopi law but can hamper an accurate measurement of its effects. To compare a stable level of American movie sales with French movie sales by town before and after the Hadopi law, theater movie sales data are aggregated by town, by month and by nationality.<sup>17</sup> This aggregation mitigates the competitive effects between theaters in a town and smooths most of the erratic variation in movie sales in each town. We estimate the following equation:

$$Outcome_{ct} = \rho_c + \tau_t + \alpha Hadopi_t * Internet_c + x'_{ct}\beta + \epsilon_{ct} \tag{1}$$

where  $Outcome_{ct}$  is an economic outcome (the logarithm of admissions to all movies and the market share of American movies) in geographic area (town) c during month t,  $Hadopi_t$  equals one after the implementation of Hadopi,  $Internet_c$  is the fraction of household connections subscribing to high-speed Internet over the total number of connections in town c during the first quarter of 2009, and  $x_{ct}$  is a set of time-varying covariates we describe below. The standard errors are clustered at the town level, as recommended by Bertrand et al. (2004), to take into account the autocorrelation between observations belonging to the same town.

The local broadband Internet use rate is potentially endogenous because it is a choice variable. For instance, households may have decided to change their subscription to broadband Internet to watch movies online rather than in the theater. We obtain exogenous variation in Internet use as in Bhuller et al. (2013) by exploiting the spatial variation of high speed Internet infrastructures. To instrument  $Internet_{ct}$ , we estimate the following first stage equation:

$$Internet_{ct} = \delta Infractructure_c + \omega_{ct} \tag{2}$$

where  $Infrastructure_c$  is the fraction of connections with high speed Internet over the total number of connections in town c.

The infrastructure to access high-speed Internet was introduced in France with the installation of phone connections. The spatial variation in Internet infrastructure is linked to its progressive diffusion in France and is due to limited funding.

<sup>&</sup>lt;sup>17</sup>Bi-national movies such as Harry Potter are considered American.

The timing of the diffusion of high-speed Internet is not correlated with the Hadopi law and does not vary substantially over the period 2008 - 2011. The theoretical maximum download speed of an Internet connection depends on the infrastructure and is not the result of consumer choice. Internet service providers only advertise the maximum speeds they deliver in France as a whole. In principle, most consumers do not know the speed of their Internet connection before moving into their homes. However, any consumer can find this information using detailed maps available on the Internet. In remote areas, that some homes have poor Internet connections can be common knowledge. This feature might weaken the power of our instrument. A good feature of our instrument is that all Internet packages were unlimited over the period 2008 - 2011, and hence, a consumer could not choose to restrict the speed of his or her Internet connection by paying less.

As Liebowitz (2008) notes, there are at least two limitations with using the broadband Internet use rate as a proxy for online piracy. First, the Internet use rate represents the number of users, not the intensity of Internet use. However, it seems reasonable to assume that the intensity of Internet use is strongly associated with the download speed allowed by the connection. Second, the Internet can influence movie admissions in theaters: it is a channel that convey media that can compete with movie theaters.<sup>18</sup> As a consequence, the  $\alpha$  coefficient captures both the online piracy effect and the competing Internet effect. Nonetheless, our local-analysis begins in the middle of 2008, when the Internet was less of a new and growing form of alternative media. We believe that the competing effect of Internet has not experienced a drastic change coincidentally with the Hadopi law, and hence it may not be a substantial issue.

Unobservable town-level shocks and unobservable common time shocks that are correlated with movie performance are controlled for through town and through month fixed effects. We also control for a local time-varying indicator with  $x_{ct}$ , which contains factors influencing the box office performance of movies. That is, we control for the quality of all American and French movies available in town c during month t using the results of an ancillary estimation. We obtain these two movie quality indicators by estimating  $\hat{\xi}_i$  using the following equation:  $log(admissions_{icet}) = \gamma_t + \phi_c + \nu_e + \mu(t - r_i) + \xi_i + \zeta_{icet}$ , where  $log(admissions_{icet})$  is the logarithm of the admissions of movie i, in town c, in theater e, during week t, where  $\gamma_t$  is a week indicator,  $\phi_c$  is a town indicator,  $\nu_e$  is a movie theater establishment indicator,  $r_i$  is the release date of movie i, such that  $\mu$  controls for the common decay effect of movie sales, and  $\xi_i$  is a movie indicator.  $\hat{\xi}_i$  are then aggregated by month, town and nationality of movies, weighted by the number of movie theaters showing the movie i.

3D movies were introduced at essentially the same time as Hadopi with the release of Avatar near

<sup>&</sup>lt;sup>18</sup>Another influence would be that the Internet can spread electronic word of mouth.

the end of 2009. This might have influenced the effect of the anti-piracy law because this technological innovation could have increased the demand for movies in theaters and 3D films were mainly American.<sup>19</sup> The effect of 3D films might also have been different across towns because the diffusion of digital projectors in France was progressive and due to limited funding. As a consequence, we additionally control for the number of movies played in 3D in town c during month t. In this way, we also partially control for the rapid conversion to digital projectors in France, as 3D movies require digital projectors to be exhibited in theaters.

#### 5.1.2 Results

Table 8 and table 9 report the results with the market share of American movies and the log of total admissions as the outcomes of interest. Columns 1-2 control for town fixed effects and month fixed effects, columns 3-4 additionally control for the time-varying quality of American movies and French movies, columns 5-6 add the control for 3D screenings, and columns 7-8 contain all of the control variables and drop the 50% of towns with high-speed Internet infrastructure closest to the median as a robustness check to intensify the effect of Hadopi. The results indicate that American movies' market share increase by approximately 0.9 percentage point when the broadband Internet use rate increases by 1 percentage point, after the implementation of the anti-piracy law. The regressions also indicate no effect of the Hadopi law on the total number of admissions in France. These effects are robust to different specifications and do not appear before the actual implementation of the law. The results indicate that the Hadopi effect is most likely to be a pure business stealing effect that benefits U.S. movies at the expense of other (French) movies. However, the methodology requires supplementary analyses to exclude other competing explanations.

				U.S. mar	ket share			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Hadopi $\times$ Internet	0.539		$0.529^{*}$		$0.569^{*}$		$0.816^{**}$	
	(0.330)		(0.282)		(0.309)		(0.318)	
Hadopi $1 \times Internet$		0.621		0.547		0.533		0.600
		(0.520)		(0.491)		(0.493)		(0.561)
Hadopi $2 \times Internet$		0.840		0.766		0.780		0.979*
		(0.574)		(0.507)		(0.513)		(0.552)
Hadopi $3 \times$ Internet		0.915*		0.902**		0.959**		$1.363^{***}$
		(0.505)		(0.432)		(0.463)		(0.456)
Town fixed effect	У	у	У	у	У	у	У	У
Month fixed effect	У	У	У	У	У	У	У	У
Quality of French films			У	У	У	У	У	У
Quality of U.S.			У	У	У	У	У	У
3D shows					У	У	У	У
Constant	$0.541^{***}$	$0.489^{***}$	$0.394^{***}$	0.140	$0.394^{***}$	0.0949	$0.539^{***}$	$0.544^{***}$
	(0.00562)	(0.00829)	(0.00733)	(0.247)	(0.00730)	(0.276)	(0.00812)	(0.00815)
Observations	2087	2087	2087	2087	2087	2087	990	990
R-squared	0.880	0.880	0.886	0.886	0.886	0.887	0.892	0.893

Table 8: DiD local analysis based on instrumentalized high speed Internet use: U.S. market share

Standard errors clustered by town \* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01

<sup>19</sup>Avatar was followed by other 3D releases such as How to Train Your Dragon (2010), Alice in Wonderland (2010), and so forth.

log(Total Admissions) (1)3.585 (5)2.959 (7)2.351 (2)(3)(4)(6)(8)Hadopi  $\times$  Internet 2.872(2.599)(2.112)(2.158)(2.145)Hadopi $1 \times Internet$ 1.680 1.081 1 048 -0.170(1.531) $(1.545) \\ 3.154$ (1.554)(1.821)Hadopi $2 \times Internet$ 4.3153.185 1.969 (3.084)(2.496)(2.521)(2.526)Hadopi $3 \times$  Internet 4.7103.8403.9702.628(2.999)(2.522)(2.557)(2.549)Town fixed effect у y y y y у у у у y v Month fixed effect v у ν у Quality of French films y у у у v у Quality of U.S. 3D shows у у y y у

7.183\*\*\*

(1.448)

2087

0.942

9.647\*\*\*

(0.0434)

2087

0.942

9.059\*\*\*

(0.871)

2087

0.942

9.805\*\*

(0.0339)

990

0.934

9.804\*\*\*

(0.0349)

990

0.934

9.647\*\*\*

(0.0436)

2087

0.942

Table 9: DiD local analysis based on instrumentalized high speed Internet use: Total admissions

Standard errors in parentheses + p < 0.10, \* p < 0.05, \*\* p < 0.01, \*\*\* p < 0.001

9.801\*\*

208'

0.941

(0.0253)

### 5.2 Analysis 2: American vs French movies in France

9.447\*\*\*

(0.0313)

2087

0.941

#### 5.2.1 Empirical strategy

Constant

Obervations

R-squared

We check that our results are not driven by supply reactions by conducting an additional analysis at the French national level in a difference-in-differences setting. The "treatment" group comprises American movies, and the "control" group is composed of French movies. The baseline estimation is as follows:

$$Outcomes_{it} = \rho_t + \beta X_i + \gamma U S_i + \alpha Hadopi_t \times U S_i + \epsilon_{it}$$
(3)

for movie *i* during release week *t*, where  $US_i$  equals 1 if movie *i* is from the US and 0 otherwise, and where  $X_i$  is a set of movie variables including the production budget, the advertising expenditures, the number of screens on the release day, consumers' rating, press rating, fixed effects for genre, nationality, art-house movies, age restrictions, and distributors. We test four possible supply reactions to the Hadopi law: the log of the number of screens on which movies are released, the log of the production budget in dollars, the log of the total marketing expenditures, and the quality of movies measured by consumer ratings.

In addition, we also verify that the market share of U.S. movies has increased thanks to the antipiracy law. We estimate equation 3 using the logarithm of admissions, and with the logarithm of the ratio between the number of admissions during the release week and the corresponding number of screens as dependent variables. This ratio is commonly used in the industry and enables re-scaling all movies, thus making comparisons more relevant and controlling for the main supply reaction. As discussed in Section 3, American movies are much more frequently pirated online than are French movies during their release week, mainly because the pirates providing illegal digital copies are much more active in the U.S.. To the best of our knowledge, we are the first to exploit the difference in online piracy between U.S. movies and domestic movies, within a country, to estimate the effects of illegal file sharing on movie sales. This approach does not allow us to disentangle the market creation effect from the business stealing effect. It identifies relative business stealing by comparing American to French movies: American movies are directly impacted by the law, and French movies are indirectly impacted through the greater competition from American films following the law.

### 5.2.2 Results

We present the results of the supply reaction in table 10. They indicate that there was no change in the number of opening screens for U.S. movies during Hadopi implementation (columns 1 and 2), but there was a slight decrease in the production budget of American movies (columns 3 and 4), and a decrease in consumer ratings (column 5 and 6). We also observe a small increase in marketing expenditure during the third period of Hadopi (columns 7 and 8). Overall, these results exclude supply side reactions to the law as the main explanation for the increase in American movies' market share. They tend to indicate a decrease in American movie quality (the production budget and the consumer ratings) during the Hadopi effect, which actually reinforces the main result of an increase in demand for American movies. The increase in the marketing expenditure of American movies during the third period of Hadopi could have explained the rise in U.S. movies' market share, but we confirm the increase in U.S. movie sales relative to French ones, while controlling for advertising, in a robustness check presented below.<sup>20</sup> Then, this increase in advertising spending does not entail an increase in the number of screens, and hence, it might be linked to the legal reduction of the time to release a movie on video.<sup>21</sup> It could also partially come from a reaction on the part of distributors to the increased consumer demand for American films.

The results in table 11 confirm that Hadopi is associated with an increase in admissions for U.S. movies relative to French ones. They also indicate that Hadopi may have had an effect before its actual implementation during the period of legislative debate (April 2009 - October 2009). Finally, they also show that the Hadopi effect may have been temporary and ended in September 2010, the period during which the graduated response began to be enforced. However, this is the only analysis to exhibit such a

<sup>&</sup>lt;sup>20</sup>In addition to this robustness check, we verify that the efficiency of advertising has not increased simultaneously with Hadopi implementation by estimating the following equation for U.S. movies:  $log(Admissions_it) = \rho_t + \delta log(AdSpending) + \beta log(AdSpending) \times Hadopi_t + \gamma X_{it}$ , where  $X_{it}$  includes the usual control variables. It appears that  $\beta$  is not significantly different from zero.

<sup>&</sup>lt;sup>21</sup>Simultaneous with the Hadopi law, the minimum period of time to release a movie on video after the theatrical release was reduced from six to four months.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	log(Screens)	log(Screens)	log(Budget)	log(Budget)	log(Ad)	log(Ad)	Consumer	Consumer
							Ratings	Ratings
Hadopi $\times U.S.$		-0.0187		-0.207*		0.140*		-0.257***
		(0.0506)		(0.108)		(0.0734)		(0.0771)
Hadopi1 $\times U.S.$	0.102		0.188		0.0646		-0.300*	
	(0.104)		(0.203)		(0.182)		(0.123)	
Hadopi2 $\times U.S.$	0.0517		-0.304*		0.0852		-0.312**	
	(0.0657)		(0.159)		(0.0930)		(0.109)	
Hadopi $3 \times U.S.$	-0.0440		-0.0663		$0.208^{**}$		-0.303***	
	(0.0656)		(0.132)		(0.0811)		(0.0893)	
log(Budget)	$0.248^{***}$	$0.247^{***}$			0.0283	0.0303	-0.0104	-0.0102
	(0.0475)	(0.0484)			(0.0361)	(0.0355)	(0.0394)	(0.0392)
log(Ad)	$0.310^{***}$	$0.312^{***}$					$0.0672^{*}$	0.0659*
	(0.0671)	(0.0672)					(0.0327)	(0.0331)
log(Screens)					$0.886^{***}$	$0.885^{***}$	-0.0281	-0.0363
					(0.0539)	(0.0545)	(0.0633)	(0.0632)
Consumer rating	-0.0142	-0.0183	0.0176	0.0117	$0.0926^{**}$	0.0896*		
	(0.0324)	(0.0323)	(0.0771)	(0.0763)	(0.0456)	(0.0476)		
Press rating	$0.0469^{**}$	$0.0497^{**}$	0.0434	0.0427	-0.0164	-0.0169	$0.377^{***}$	$0.378^{***}$
	(0.0237)	(0.0239)	(0.0510)	(0.0507)	(0.0295)	(0.0298)	(0.0509)	(0.0513)
Genre fe	У	У	У	У	У	У	У	У
Nationality fe	У	У	У	у	У	У	У	У
Art and House fe	У	У	У	у	У	У	У	У
Age restriction fe	У	У	У	У	У	У	У	У
Month fe	У	У	У	у	У	У	У	У
Distributor fe	У	У	У	у	У	У	У	У
Ν	604	604	646	646	767	767	604	604
$R^2$	0.728	0.726	0.749	0.747	0.741	0.740	0.567	0.562
Robust standa	d orrors							

Table 10: DiD French vs. American in France (first week), supply reaction: number of opening screens, production budgets, advertising expenditures, and consumer ratings

\* p < 0.10, \*\* p < 0.05, \*\*\* p < 0.01

temporary effect beginning before the implementation of the law. We further implement several robustness tests to check these results.

Our first robustness check concerns extreme values. A few movies can capture nearly the entire audience, and the estimated Hadopi effects may simply be driven by few outstanding American (French) movies that coincidently performed well (poorly) during the Hadopi period. With a two-step procedure, we first estimate the main specifications and then re-estimate the same specifications having dropped the 20% of movies with the highest absolute residual values from the first estimate. The results are consistent with those obtained from the full sample; the main results are even reinforced with this procedure and are presented in Appendix D in table 16.

Thus far, we have assumed the effect of Hadopi to be the same for all American movies. However, the effect of the Hadopi law could be heterogeneous across American movies because online piracy is likely to affect more popular movies. Thus, as a second robustness check, we estimate the quantile treatment effect to determine the heterogeneous effect of the Hadopi law on movies box office performance.<sup>22</sup> Others

 $<sup>^{22}</sup>$ We have extended this robustness check to the supply side variables. There is no noteworthy increase in the advertising expenditure for American movies during the Hadopi period into the right tail of the advertising expenditure distribution and not much into the rest of the distribution either. We also see no effect of the Hadopi law on the distribution of the number of opening screens, and of the production budget of American movies. Lastly, the decrease in consumer ratings for American movies during the Hadopi period seems to concern the whole distribution. All results can be obtained upon request, they are not presented for saving space.

advantages of quantile regression are its robustness to outliers and its robustness when the dependent variable has a highly non-normal distribution (see Koenker (2005)), which is important because outliers are not unusual in the movie industry: a few winners take all of the revenues. The results suggest that the positive effect of the Hadopi law on American movies' box office performance becomes larger as we move upward into the right tail of the distribution. In other words, the effect of Hadopi is stronger on the most successful movies, which are the most pirated ones (see table 17 in the Appendix). This finding confirms that online piracy reduces the superstar phenomenon (Rosen (1981) and Adler (1985)) as in Gopal et al. (2006). We find no negative effect of the Hadopi law in the left tail of the distribution, and hence, high-performing American movies do not benefit from the Hadopi law, to the detriment of less well performing American movies.

-			(	)
	(1)	(2)	(3)	(4)
	log(Admissions)	log(Admissions)	log(Admissions/Screens)	log(Admissions/Screens)
Hadopi $\times U.S.$	0.110 +		0.103	
	(0.0657)		(0.0680)	
Hadopi1 $\times U.S.$		0.265*		0.302*
		(0.123)		(0.126)
Hadopi2 $\times U.S.$		0.239**		0.257**
		(0.0905)		(0.0945)
Hadopi $3 \times U.S.$		0.0952		0.0798
		(0.0766)		(0.0800)
log(Screens)	$1.393^{***}$	1.382***		
	(0.0759)	(0.0741)		
log(Ad)	0.101*	0.101*	0.231***	0.228***
	(0.0434)	(0.0418)	(0.0511)	(0.0491)
log(Budget)	-0.0292	-0.0265	0.0674	0.0675
	(0.0407)	(0.0400)	(0.0428)	(0.0414)
Consumer rating	0.198***	0.206***	0.192***	0.201***
-	(0.0425)	(0.0425)	(0.0445)	(0.0442)
Press rating	0.120***	0.116***	0.132***	0.127***
	(0.0308)	(0.0306)	(0.0312)	(0.0310)
Genre fe	у	y	у	У
Nationality fe	У	У	У	У
Art and House fe	y	y	y	y
Age restriction fe	У	У	У	У
Month fe	У	У	У	У
Distributor fe	У	У	У	У
Constant	$2.713^{***}$	$2.764^{***}$	3.770***	3.793***
	(0.350)	(0.348)	(0.391)	(0.379)
Ν	767	767	767	767
$R^2$	0.801	0.803	0.444	0.452

Table 11: DiD French vs. American movies in France (first week) : demand reaction

Robust standard errors ; Consumer and Press ratings come from Allocine. fr + p < 0.10, \* p < 0.05, \*\* p < 0.01, \*\*\* p < 0.001

5.3 Analysis 3: International comparisons

#### 5.3.1 Empirical strategy

A potential explanation for the better performance of American movies is that Hadopi triggered a supplyside reaction by U.S. majors. We began to rule out this explanation in the previous section by comparing the supply of American movies to the supply of French movies. However, it is possible that U.S. majors decided to open their movies on more screens in France, as a response to the law, relative to their strategy in other European countries. To further assess the positive effects of the Hadopi law on American movies' box office performance, we compare some outcomes relative to the supply level of American movies between France and a control group composed of different European countries at the movie level.

The estimation equation is the following one :

$$Outcome_{ict} = \rho_c + \rho_t + \beta X_i + \alpha Hadopi_{ct} + \epsilon_{ict}$$

$$\tag{4}$$

where  $X_i$  a set of movie variables that are defined as above and  $Hadopi_{ct}$  is a post-Hadopi dummy variable that equals one only for France. Our dependent variables are the natural logarithm of the ratio between the number of admissions for movie *i* during first week *t* in country *c*,  $Admissions_{ict}$ , and the corresponding number of screens  $Screens_{ict}$ . We also directly test two possible supply reactions to the law: the quantity of U.S. movies exported to country *c*, measured by the log of the number of opening screens of exported movies, and the speed of exports, measured by the time lag in days between the date of release in the U.S. and the foreign release date (McCalman (2005)).

As a check of the main the effects of the Hadopi law on the demand, we compare France with other European countries with some aggregate outcomes of movie performance. To detect a total demand creation effect and a redistributive effect between American movies and domestic movies, we aggregate movie-level admissions data by country and by month. We use the following simple specification with data at the country  $\times$  month level (see Adermon and Liang (2014) for a similar approach):

$$Outcome_{ct} = \rho_c + \rho_t + \alpha Hadopi_t \times France_c + \epsilon_{ct}$$
<sup>(5)</sup>

where c is a country index and t a time period index.  $Hadopi_t$  is a post-Hadopi dummy variable equal to one after November 2009. Finally,  $\rho_c$  is a country fixed effect,  $\rho_t$  a quarterly fixed effect, and  $\epsilon_{ct}$  an idiosyncratic error term. We study the effect on two aggregate outcomes: (1) the log of the total number of admissions and (2) the log of American movies market shares.

An important issue for the identification strategy is to have a convincing control group (Meyer (1995)). During the period 2008-2011, no measure similar to the French Hadopi law was implemented in Europe other than a 2009 Swedish law, and hence Sweden is always excluded from the control group (see Appendix E for a discussion). Then, to select the countries to be included in the control group, we use a methodology that ensures that the pre-treatment trends in the dependent variables of interest are similar to the trend in the treated country. We first visually check that there is no divergence in trends between France and a candidate control country. Then, we regress the variable of interest in France on this variable in the candidate control country for the months preceding the beginning of the public debate on Hadopi (April 2009) to select the candidates that are significantly correlated with France, as proposed by Hilger et al. (2011). Details on the construction of the control group are available in Appendix E.

#### 5.3.2 Results

The results from equation 4 are presented in Table 12. They indicate that American movies performed substantially better in France than in other countries after the implementation of Hadopi (columns 1 and 2). There is, however, no change in the speed of their exports (columns 3 and 4) or in the number of opening screens (columns 5 and 6), although we note that Hadopi3 is significantly negative in column 4. This decrease in the number of opening screens for American movies in France actually strengthens our main result. The time lag between the release in the US and in France increases during Hadopi1 but not during Hadopi2 or  $Hadopi3.^{23}$  Overall, this analysis confirms that a supply reaction is not the main explanation for the positive effect of the Hadopi law on American movies.

Table 13 provides the estimation results from equation 5. They indicate no effect of the Hadopi law on the total number of admissions in France (columns 1 and 2). These regressions also indicate that American movies' market share increased by approximately 8% in France thanks to Hadopi relative to the control group (columns 3 and 4). This effect seems permanent and does not appear before the actual implementation of the law. This difference-in-differences strategy confirms that there is no demand creation following the law and that the mechanism at play is most likely a pure substitution effect to the benefit of American movies.

### 5.4 Analysis 4: Consumer-level approach

In this section, we provide some evidence to better understand the consumer behavior underlying the redistributive effect that benefits American movies.

Consumers who download more should be more sensitive to the anti-piracy law. For such consumers, we should observe a more important shift in their legal consumption of American movies in theatres. Using consumer-level data, our last approach provides additional evidence supporting this mechanism using the difference in characteristics (age) between consumers who have unlimited access card.

The benefit of this data collection approach is that it provides us with information on the type of consumers who purchase movies in theaters. Because we could only collect those data for two middle-size

 $<sup>^{23}</sup>$ The Hadopi law has also reduced the legal delay between the release date of movies in theaters and the video release date from 6 to 4 months, which may explain this increase in the time lag in France.

	$log(\frac{Admissions}{2})$	log( <u>Admissions</u> )	log(Screens)	log(Screens)	Lag	Lag
	(1)	(2)	(3)	(4)	(5)	(6)
	( )	( )	( )	( )	( )	
Hadopi × France	0.217***		-0.132***		0.292	
Hudopi // Hulleo	(0.0419)		(0.0429)		(0.795)	
Hadopi $1 \times$ France	()	0.0599	()	-0.0977	()	$2.143^{*}$
I I I I I I I I I I I I I I I I I I I		(0.0682)		(0.0621)		(1.270)
Hadopi $2 \times$ France		0.224***		-0.0651		-0.106
1		(0.0737)		(0.0677)		(1.459)
Hadopi $3 \times$ France		0.230***		-0.195***		1.129
-		(0.0455)		(0.0492)		(0.879)
Log(Budget)	$0.240^{***}$	0.240***	$0.424^{***}$	0.425***	-1.444	-1.442
	(0.0314)	(0.0314)	(0.0307)	(0.0307)	(1.198)	(1.199)
Consumer rating	$0.354^{***}$	$0.354^{***}$	0.129***	0.128***	0.637	0.652
	(0.0528)	(0.0528)	(0.0453)	(0.0451)	(1.123)	(1.122)
Press rating	-0.0341	-0.0341	0.00939	0.00879	0.0194	0.0251
	(0.0337)	(0.0337)	(0.0321)	(0.0320)	(0.580)	(0.576)
Comp. age top 5	$0.0370^{***}$	$0.0371^{***}$	0.00786	0.00888	0.265	0.265
	(0.0123)	(0.0123)	(0.0172)	(0.0173)	(0.329)	(0.331)
Comp. perc. new screens	-1.025***	-1.025***	-0.896***	$-0.894^{***}$	6.800	6.899
	(0.181)	(0.181)	(0.217)	(0.217)	(5.253)	(5.329)
Genre FE	У	У	У	У	У	У
Nationality FE	У	У	У	У	У	У
Art and House FE	У	У	У	У	У	У
Age restriction FE	У	У	У	У	У	У
Month FE	У	У	У	У	У	У
Constant	$4.956^{***}$	4.947***	$1.839^{***}$	$1.829^{***}$	$11.00^{***}$	$10.63^{***}$
	(0.235)	(0.236)	(0.238)	(0.239)	(3.970)	(3.961)
Observations	3228	3228	1254	1254	1605	1605
R-squared	0.393	0.393	0.870	0.870	0.157	0.158

Table 12: American movies across countries

Robust standard errors clustered by movie; Consumer and Press ratings come from Allocine.fr; \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1; Control Group for log(Admissions/Screens): Belgium, Finland, Germany, the Netherlands, Spain, Switzerland, and United Kingdom; Control Group for log(Screens): Belgium, and the Netherlands; Control Group for Lag: Belgium, Italy, and United Kingdom.

### Table 13: DiD International comparison

	Table 19. Did international comparison						
	(1) log(Admissions)	(2)	(3)	(4)			
	log(Admissions)	log(Admissions)	$\log(0.5. \text{ market share})$	$\log(0.5. \text{ market share})$			
Hadopi $\times$ France	0.0833		$0.0805^{***}$				
	(0.0529)		(0.0209)				
Hadopi $1 \times$ France		-0.0211		0.00190			
		(0.0238)		(0.0339)			
Hadopi $2 \times$ France		0.0821		$0.0894^{***}$			
		(0.0449)		(0.0153)			
Hadopi 3 $\times$ France		0.0770		$0.0756^{**}$			
		(0.0655)		(0.0239)			
Country fe	У	У	У	У			
Month fe	У	У	У	У			
Constant	$14.04^{***}$	$14.04^{***}$	-0.252***	-0.251***			
	(0.0356)	(0.0360)	(0.0340)	(0.0341)			
Observations	420	420	420	420			
R-squared	0.978	0.978	0.677	0.677			

Standard error are clustered by country allowing for observations to be correlated within a country. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1; Control Group for  $\log(Admissions)$ : Belgium, Germany, the Netherlands, Spain, Switzerland, and United Kingdom; Control Group for  $\log(U.S.$  market share): Belgium, Germany, Italy, Spain, Switzerland, and United Kingdom.

cities in France, the results in this section are not nationally representative. The effect in France as a whole could admittedly be different from that we describe hereafter.

#### 5.4.1 Empirical strategy

Young consumers download illegally more often than old consumers (see Hadopi (2011)), and hence they are more likely to be affected by Hadopi. We use the information on the age of consumers with unlimited access cards to investigate which types of consumers were influenced by the anti-piracy law.<sup>24</sup>

This strategy exploits a difference in illegal downloading between young and older people. We implicitly assume that consumers with cards illegally download movies despite having unlimited access to movies in theaters. Such behavior might be due to a time constraint. Indeed, the opportunity cost of watching a movie in a theater is higher than that of watching a movie at home, even if the experience of consuming a movie on a computer or in a theater is clearly different. The costs of commuting and of waiting at the theater are the main components of this time cost. Further, it is easier to stop watching a movie that one does not enjoy at home than in a movie theater. Another explanation would be a preference for watching movies as soon as they are available, on the Internet, even if they will subsequently be available in movie theaters.

We aggregate individual transaction data from consumers with unlimited access cards by movie, week, theater and three different age groups: consumers between 10 and 29 years old, between 30 and 49 years old, and between 50 and 80 years old. The equation we estimate is as follows:

$$Sales_{ijta} = \delta_i + \delta_j + \delta_t + \phi NumberCards_{jta} + \sum_{\tau=1}^{14} \beta_\tau \mathbb{1}_{\{t-r_i=\tau\}} + \sum_{a=1}^2 \gamma_a Age_a + \sum_{a=1}^2 \lambda_a USA_i * Age_a + \sum_{a=1}^2 \eta_a Hadopi_t * Age_a + \mu Hadopi_t * USA_i + \sum_{a=1}^2 \beta_a Hadopi_t * USA_i * Age_a + \epsilon_{ijta}$$
(6)

where the left-hand-side variable is the log of sales (number of admissions) from consumers with unlimited access cards belonging to age group a for movie i, during week t, in theater j.<sup>25</sup> Age<sub>a</sub> is an indicator variable that equals one for sales associated with consumers who belong to age group a. Numbercards<sub>jta</sub> is the number of consumers belonging to age group a who have used an unlimited card at least once during month t at theater j. This variable accounts for the economic success of the loyalty

 $<sup>^{24}</sup>$ We also tested for a difference between women and men, but no significant difference was found.

<sup>&</sup>lt;sup>25</sup>When a = 1, this corresponds to people aged between 10 and 29, a = 2 being people aged between 30 and 49, and a = 3 matching people aged between 50 and 80.

program of the cinema chain and particularly the large increase in the number of young cardholders. The coefficients of interest are now  $\beta_1$  and  $\beta_2$ . They measure the increase in the sales of American movies relative to other movies after the Hadopi law was passed from consumers with cards, separately for those aged between 10-29 years and between 30-49 years, in comparison to the increase by consumers with cards aged between 50-80 years.  $\eta_1$  and  $\eta_2$  are also interesting because they indicate whether the sales of French films among younger consumers increased after the law relative to consumption by older consumers. A detailed discussion of the interpretation of the  $\beta_a$  coefficients is provided in Appendix F.

#### 5.4.2 Results

The columns 1 to 4 of Table 14 report the estimation results for equation (6). We observe an increase in American movies sales for younger consumers, relative to older ones, associated with Hadopi enforcement. The increase in the consumption of American movies is decreasing with age. Compared with consumers between 50 and 80 years of age, and compared with the sales of French movies, the increase in American movie sales after the anti-piracy law was passed is approximately 18% (statistically significant at the 1% level) for consumers between 10 and 29 years old and is twice the value of the increase for consumers between 30 and 49 years old, which is approximately 9%. This is consistent with illegal media consumption on the Internet decreasing with age (see Hadopi (2011)). The coefficients  $\eta_1$  and  $\eta_2$  are either negative or not significantly different from zero and indicate that sales of French movies to younger consumers decreased after the introduction of the Hadopi law. It confirms the substitution effect of the anti-piracy law to the benefit of American movies. Finally, this analysis reveals an effect beginning before the adoption of the law.

As a robustness check, we estimate equation (6) that controls for the increase in the number of consumers subscribed to the unlimited pass by age group. The results are presented in columns 5 to 8. The findings are similar: the coefficient  $\beta_1$  is still positively significant and larger than  $\beta_2$ , which is also still positive. The main difference lies in the coefficients  $\eta_1$  and  $\eta_2$ , which are more significantly negative than in columns 1 to 4. Taking into account the increase in the number of consumers subscribing to an unlimited card strengthens the redistributive effect that the increase in U.S. market share comes at the expense of French movies.

different age groups								
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	stand. err.	stand. err.	stand. err.	stand. err.	stand. err.	stand. err.	stand. err.	stand. err.
	clustered	clustered	clustered	clustered	clustered	clustered	clustered	clustered
	by movie	by movie	by movie	by movie	by movie	by movie	by movie	by movie
	and		and		and		and	
	by theater		by theater		by theater		by theater	
Hadopi $\times$ Age10-29	0.108 +	0.108			-0.125*	-0.125		
	(0.0553)	(0.0799)			(0.0616)	(0.0854)		
Hadopi $\times$ Age30-49	$-0.129^{***}$	-0.129**			$-0.148^{***}$	-0.148**		
	(0.0330)	(0.0468)			(0.0335)	(0.0466)		
Hadopi × USA × Age10-29	$0.189^{**}$	$0.189^{*}$			$0.192^{**}$	0.192*		
	(0.0674)	(0.0948)			(0.0668)	(0.0938)		
Hadopi $\times$ USA $\times$ Age30-49	0.0994*	0.0994			$0.103^{*}$	0.103 +		
	(0.0438)	(0.0619)			(0.0443)	(0.0616)		
Hadopi1 $\times$ Age10-29			-0.0569	-0.0569			-0.102	-0.102
			(0.0858)	(0.123)			(0.0861)	(0.123)
Hadopi1 $\times$ Age30-49			-0.0406	-0.0406			-0.0558	-0.0558
			(0.0495)	(0.0702)			(0.0505)	(0.0699)
Hadopi2 $\times$ Age10-29			0.0945	0.0945			-0.0694	-0.0694
			(0.0783)	(0.112)			(0.0797)	(0.113)
Hadopi2 $\times$ Age30-49			-0.0798 +	-0.0798			-0.103*	-0.103
			(0.0464)	(0.0651)			(0.0471)	(0.0650)
Hadopi $3 \times \text{Age10-29}$			0.0877	0.0877			-0.253**	-0.253*
			(0.0686)	(0.0994)			(0.0808)	(0.106)
Hadopi $3 \times \text{Age}30-49$			-0.178***	-0.178**			-0.205***	-0.205***
			(0.0420)	(0.0595)			(0.0427)	(0.0590)
Hadopi1 $\times$ USA $\times$ Age10-29			0.292**	0.292*			0.289**	0.289*
			(0.104)	(0.143)			(0.104)	(0.143)
Hadopi1 $\times$ USA $\times$ Age30-49			0.188* <sup>*</sup>	0.188*			0.182* <sup>*</sup>	0.182 +
			(0.0675)	(0.0942)			(0.0690)	(0.0939)
Hadopi $2 \times \text{USA} \times \text{Age10-29}$			0.202*	0.202			0.198*	0.198
			(0.0962)	(0.134)			(0.0945)	(0.133)
Hadopi $2 \times \text{USA} \times \text{Age}30\text{-}49$			0.133*	0.133			0.132*	0.132
. 0			(0.0632)	(0.0896)			(0.0640)	(0.0892)
Hadopi $3 \times \text{USA} \times \text{Age10-29}$			0.323***	0.323**			0.337***	0.337**
. 0			(0.0832)	(0.118)			(0.0826)	(0.116)
Hadopi $3 \times \text{USA} \times \text{Age}30\text{-}49$			0.178**	0.178*			0.182***	$0.182^{*}$
. 0			(0.0544)	(0.0767)			(0.0551)	(0.0762)
Number of unique members			()	()	у	у	y	y
N	21522	21522	21522	21522	21522	21522	21522	21522
R2	0.811	0.811	0.812	0.812	0.813	0.813	0.814	0.814
	.1							

Table 14: Triple-differences on consumers with unlimited access card: a comparison between different age groups

 $\begin{array}{l} \mbox{Standard errors in parentheses} \\ + \ p < 0.10, \ * \ p < 0.05, \ ** \ p < 0.01, \ *** \ p < 0.001 \end{array}$ 

### 6 Welfare implications and conclusion

### 6.1 Welfare implications

The four empirical approaches all lead to the conclusion that the market share of American movie sales increased after the advent of the anti-piracy law. The estimated values of the coefficients are different, but they have similar implications after some adjustments are made to compute the total change in the number of admissions to U.S. films in France after the law. The implied effects are shown in Table 15. The number of admissions is different before and after the law, but we are only interested in the differences that are attributable to the law, and our estimation results show that there is no sizable demand creation caused by the law. Therefore, to compute the effects on the market share of American movie sales, we first assume that the total number of admissions was the same before the law.

Then, we note that the U.S. films' market share was equal to 48% after the Hadopi law was passed and that the total number of admissions over the post-Hadopi period was approximately 440 million.<sup>26</sup> Finally, we use the distribution of broadband Internet use presented in table 3 to compute the effect from the town-level approach and the consumer age distribution to compute the effect from the consumer-level approach.<sup>27</sup>

10010 103	· comparison or the	results and variation	in revenue and wen	are
	Study 1: Town level approach	Study 2: American vs French movies in France	Study 3: International comparisons	Study 4: Consumer level approach
Estimated coefficient	Increase in the U.S. market share by 0.6 percentage point when the high speed Internet use rate increase by 1 percentage point	Increase in the number of admissions to U.S. films by 10%	Increase in the U.S. market share by 8%	Increase in the number of admissions to U.S. films for consumers aged 10-29 (resp. 30-49) by 20% (resp. 10%)
Rise in the market share of U.S. films	9%	10%	8%	10%
Rise in the number of admissions to U.S. films (in million)	17.1	19.2	15.4	20.0
Rise in the revenue to U.S. films $\equiv$ Decrease in the consumer welfare (in euros million)	107	121	97	126

Table 15: Comparison of the results and variation in revenue and welfare

With an increase in the U.S. market share of approximately 9%, the increase in U.S. film admissions after the law is between 15.4 and 20.0 million. It involves a decrease in admissions to French films of

<sup>&</sup>lt;sup>26</sup>For instance, with a rise in the market share of US films equal to 12.3%, we obtain a rise in the number of admissions to US films equal to 22.9 million  $(0.48 \times 440 - (0.48 \times 440/1.123) = 22.9)$ 

 $<sup>^{27}</sup>$ The distribution of moviegoers' age in France is given by the CNC and is as follows: 42.8% are aged between 6 and 29, 25.8% are aged between 30 and 49, and 31.4% are aged 50 and over.

approximately the same amount. Given that the average ticket price was 6.3 euros in France during the post-Hadopi period, U.S. movies benefited from an increase in sales of between 97 and 126 million euros. On average, a distributor collects 38% of ticket revenue. As a consequence, the transfer in revenue from French movie distributors to U.S. distributors is between 37 and 48 million euros.

There is also a loss of consumers surplus associated with the law because according to our main mechanism behind the effects of the law, some consumers watch less non-U.S.movies legally in theaters. A back-of-the-envelop calculation consumer surplus loss is given by the quantity of non-U.S. (French) movies they stop watching legally times the price of a ticket in theaters. This amount also corresponds the transfer of sales between French and U.S. movies (97 and 126 million euros). Because the marginal cost to distribute movies is almost zero, the loss in consumer surplus is also a loss of welfare (See Appendix F for a more formal discussion).

This computation is very rough and only takes into account short run effects. For instance, we can only estimate the change in welfare for consumers involved in the business stealing effect, and we do not estimate the welfare variation for users who stopped downloading movies but who were not converted into legal consumers because of a lack of data.<sup>28</sup> Finally, our estimation of welfare loss says nothing about how consumers use their new free time. In particular, we do not consider subsequent substitution with DVDs or Blu-rays, as we have no detailed information on the video market.

#### 6.2 Conclusion

In this paper, we have analyzed the effects of online piracy on movies' box office performance. We use the French anti-piracy law known as Hadopi as an exogenous shock to online piracy. Our four empirical analyses converge on the same conclusions. The Hadopi law clearly benefited American movies at the expense of French ones and created no additional demand for movies in theaters. These two effects are not driven by a supply reaction by distributors and producers.

Online piracy in a foreign country seems to distort the distribution of revenues between U.S. movies and domestic movies in favor of domestic movies. In that sense, implementing a regulation that deters illegal downloads re-establishes a 'fair' allocation of sales between U.S. and home movies according to their value to consumers. The French anti-piracy laws seem to be effective, at least partially and temporarily, in combatting file sharing. Indeed, our results are consistent with a decrease in the level of online piracy of cultural products. The long-term effect of these laws is more questionable, as the technologies associated with illegal sharing evolve quickly.

<sup>&</sup>lt;sup>28</sup>Such an exercise would require at least some data on the downloads before and after the law.

The absence of demand creation also informs us of the partial inefficiency of the law concerning the users that download movies on the Internet but that never buy them. As they are not converted into legal consumers, this finding might indicate that the law does not reduce their level of online piracy because they know how to circumvent the graduated response. It could also mean that the law reduces the online piracy level of these users while they do not begin purchasing movies because their valuation of pirated products is too low: they would have not legally consumed movies without online piracy.

Finally, the effect of an anti-piracy law is likely to have larger effects on the video market than on the theatrical market, for at least two reasons. Firstly, the supply of illegal copies of movies during the video release is dramatically higher from this supply during the theatrical release. A video release triggers the supply of many illegal copies, including domestic and American movies. Secondly, legal videos are much more substitutable with their pirated versions. It would be interesting to carry out an evaluation of the Hadopi law's effect on the video market.

### References

- A. Adermon and C.Y. Liang. Piracy and music sales: The effects of an anti-piracy law. Journal of Economic Behavior and Organization, 105:90–106, 2014.
- M. Adler. Stardom and talent. The American Economic Review, 75(1):208–212, 1985.
- F. Alcala and M. Gonzalez-Maestre. Copying, superstars and artisitic creation. Information Economics and Policy, 22:365–378, 2010.
- J. Bai and J. Waldfogel. Movie piracy and sales displacement in a sample of chinese college students. Information Economics and Policy, 24(3):187–196, 2012.
- Y. Bakos, E. Brynjolfsson, and D. Lichtlman. Share information goods. Journal of Law and Economics, 42(1):117–156, 1999.
- P. Belleflamme and M Peitz. Digital piracy: Theory. The Oxford Handbook of the Digital Economy, 2012.
- M. Bertrand, E. Duflo, and S. Mullainathan. How much should we trust differences-in- differences estimates? Quarterly Journal of Economics, 119(1):249–275, 2004.
- S. Bhattacharjee, D. Gopal, R., K. Lertwachara, and R. Marsden, J. Impact of legal threat on online music sharing activity: An analysis of music industry legal actions. *Journal of Law and Economics*, 49 (1):91–114, 2006.

- S. Bhattacharjee, R. Gopal, K. Lertwachara, J. Marden, and R. Telang. The effects of digital sharing technologies on music markets: A survival analysis of albums on ranking charts. *Manangement Science*, 53(9):1359–1374, 2007.
- M. Bhuller, T. Havnes, E. Leuven, and M. Mogstad. Broadband internet: An information superhighway to sex crime? *Review of Economic Studies*, 80(4):1237–1266, 2013.
- R. Conner, K. and P. Rumelt, R. Software piracy: An analysis of protection strategies. Management Science, 37(2):125–139, 1991.
- B. Danaher and M. Smith. Gone in 60 seconds: The impact of the megaupload shutdown on movie sales. Working paper, 2013.
- B. Danaher and J. Waldfogel. Reel piracy: the effect of online film piracy of international box office sales. Working Paper, 2012.
- B. Danaher, S. Dhanaobhon, M.D. Smith, and R. Telang. Converting pirates without cannabilizing purchasers: The impact of digital ditribution on physical sales and internet piracy. *Marketing Science*, 29 (6):125–139, 2010.
- B. Danaher, M.D. Smith, and S. Telang, R.and Chen. The effect of graduated response anti-piracy laws on music sales:evidence from an event study in france. *Journal of Industrial Economics*, 62(3):541–553, 2014.
- E. Darmon, S. Dejean, and T. Pénard. La réponse graduée de l'hadopi a-t-elle un effet sur le piratage du musique et des films ? Une étude empirique des pratiques de consommation en ligne. Working Paper, pages 1–21, 2014.
- A.S. De Vany and W.D. Walls. Estimating the effects of movie piracy on box-office revenue. *Review of Industrial Organization*, 30(3):291–301, 2007.
- I. Ehrlich. Participation in illegitimate activites: A theoretical and empirical investigation. Journal of Political Economy, 81:521–565, 1973.
- L. Einav and B. Orbach. Uniform prices for differentiated goods: The case of the movie-theater industry. International Review of Law and Economics, 27(2):129–153, 2007.
- R. Galbreth, M., B. Ghosh, and M. Shor. Social sharing of information goods: Implications for pricing and profit. *Marteking Science*, 31(4):603–620, 2012.

- M. Gentzkow. Valuing new goods in a model with complementarity: Online newspapers. American Economic Review, 97(3):713–744, 2007.
- M. Giorcelli and P. Moser. Copyright and creativity: Evidence from italian operas. Stanford Working Paper, 2015.
- D. Gopal, R., S. Bhattacharjee, and L. Sanders, G. Do artists benefit from online music sharing? *Journal of Business*, 79(3):1503–1533, 2006.
- Hadopi. Hadopi, biens culturels et usages d'internet : pratiques et perceptions des internautes français. Report, 2011.
- R. Harbauch and R. Khemka. Does copyright enforcement encourage piracy? The Journal of Industrial Economics, 58(2):306–323, 2010.
- S. Hasting, J. and M. Shapiro, J. Fungibility and consumer choice: Evidence from commodity price shocks. The Quarterly Journal of Economics, 128(4):1449–1498, 2013.
- C. Heath and B. Soll, J. Mental budgeting and consumer decisions. The Journal of Consumer Research, 23(1):40–52, 1996.
- M. Heineke, J. Economic Models of Criminal Behavior. New York: Elsevier North-Holland, 1978.
- T. Hennig-Thurau, V. Henning, and H. Sattler. Consumer file sharing of motion pictures. Journal of Marketing, 71(4):1–18, 2007.
- J. Hilger, G. Rafert, and S. Villas-Boas. Expert opinion and the demand for experience goods: An experimental approach in the retail wine market. *Review of Economics and Statistics*, 93(4):1289–1296, 2011.
- S. Jain. Digital piracy: A competitive analysis. Marketing Science, 27(4):610–626, 2008.
- S. Liebowitz. File-sharing: Creative destruction or just plain destruction? Journal of Law and Economics, 49(1):1–28, 2006.
- S. Liebowitz. Testing file-having's impact by examining record sales in cities. *Management Science*, 54(4): 852–859, 2008.
- S. Liebowitz and A. Zentner. Clash of the titans: Does internet use reduce television viewing? *The Review* of *Economics and Statistics*, 94(1):234–245, 2012.

- P. McCalman. International diffusion and intelectual property rights: An empirical analysis. Journal of International Economics, 67(2):353–372, 2005.
- B.D. Meyer. Natural and quasi-experiments in economics. *Journal of Business and Economic Statistics*, 13(2):151–161, 1995.
- F. Oberholzer-Gee and K. Strumpf. The effect of file sharing on record sales: An empirical analysis. Journal of Political Economy, 115(1):1–42, 2007.
- F. Oberholzer-Gee and K. Strumpf. File-sharing and copyright. Innovation Policy and the Economy, 10 (1):19–55, 2010.
- M. Peitz and P. Waelbroeck. Why the music industry may gain from free downloading the role of sampling. International Journal of Industrial Organization, 24(5):907–913, 2006.
- C. Peukert, J. Claussen, and T. Kretschmer. Piracy and movie revenues: Evidence from megaupload a tale of the long tail? *Working paper*, 2013.
- R. Rob and J. Waldfogel. Piracy on the high c s: Music downloading, sales displacement, and social welfare in a sample of college students. *The Journal of Law and Economics*, 49(1):29–62, 2006.
- R. Rob and J. Waldfogel. Piracy on the silver screen. Journal of Industrial Economics, 55(3):379–393, 2007.
- S. Rosen. The economics of superstars. The American Economic Review, 71(5):845-858, 1981.
- Sandvine. Global Internet Phenomena Report. Report, 2011.
- M. Smith and R. Telang. Assessing the academic literature regarding the impact of media piracy on sales. Working Paper, 2012.
- N. Takeyama, L. The welfare implications of unauthorized reproduction of intellectual property in the presence of demand network externalities. *The Journal of Industrial Economics*, 42(2):155–166, 1994.
- N. Takeyama, L. The intertemporal consequences of unauthorized reproduction of intellectual property. Journal of Law and Economics, 40(2):511–522, 1997.
- R. Varian, H. Buying, sharing and renting information goods. The Journal of Industrial Economics, 48 (4):473–488, 2000.

- A. Vernik, D., D. Purohit, and S. Desai, P. Music downloads and the flip side of digital rights management. *Marteking Science*, 30(6):1011–1027, 2011.
- J. Waldfogel. Digital piracy: Empirics. The Oxford Handbook of the Digital Economy, 2012a.
- J. Waldfogel. Copyright research in the digital age: Moving from piracy to the supply of new product. American Economic Review, 102(3):337–342, 2012b.
- A. Zentner. Measuring the effect of file sharing on music purchases. *The Journal of Law and Economics.*, 49(1):63–90, 2006.
- A. Zentner. Measuring the impact of file sharing on the movie industry: An empirical analysis using a panel of countries. *Working Paper*, pages 1–35, 2010.
- A. Zentner, D. Smith, M., and C. Kaya. How video rental patterns change as consumers move online. Management Science, 59(11):2622–2634, 2012.

### A A business stealing effect is plausible: additional proofs

The results of this paper indicate no demand creation associated with the Hadopi law, so that the increase in American market share mainly stems from cannibalization at the expense of other movies. The mechanism involves a decrease in illegal sharing of movies, a binding budget and time constraint, some consumers enjoying both U.S. and French movies, and a general preference for U.S. movies.

Many sources agree on a fall of peer-to-peer use throughout 2010 and 2011, when the graduated response was implemented in France. Mediametrie/NetRatings, a service providing measurement of online audience, reported no substantial transfer on streaming and direct download services (see graph 4).



Figure 4: The stability in usage patterns of streaming and direct download services in France.

A survey conducted by *Observatoire de la satisfaction*, a company specialized in the French broadcasting market, shows that the American movies' audience distribution and the French movies' audience distribution overlap significantly in terms of gender, age, and satisfaction rate (see graphs 5(a), 5(b), and 5(c)). They use a sample of 598 movies widely released in France between January of 2006 and November of 2009. As a consequence, a fraction of movie audience is likely to enjoy both French and U.S. movies.

Data from the French national accounts on households consumption confirm that the share of consumption of movies in theaters in the shopping basket of the average French household has been steady since 1990, fluctuating between 0.11% and 0.14% of total budget (see graph 6). As a consequence, the budget and the time constraints might constitute important limitations to the expansion of the market of theatrical exhibition.

Consumers might prefer watching U.S. movies rather than French ones in theaters. Indeed, the average production budget is tremendously higher for U.S. movies (\$63 millions) than for French ones (\$9 millions).



Figure 5: The audiences of U.S. and French movies overlap

Figure 6: The share of consumption of movies in theaters over total consumption of household is stable.



This difference reflects for instance a more famous casting or better special effects associated. The audience may also anticipate that U.S. films are superior because marketing expenditures are also higher (\$2 millions for U.S. movies against \$1.1 millions for French ones). The observed ex post satisfaction of consumers is higher for U.S. movies. Indeed, the average user rating on www.allocine.fr, the major French content aggregator on movies (www.allocine.fr), is 3.1 for U.S. films when it is 2.8 for French films. Finally, during their release week, U.S. movies are played on 360 screens on average in France when French movies are played on 320 screens. Note that these figures correspond to the sample we use in section 5, that is biased toward big French movies. For instante, French movies are released on 180 screens on average in France, when considering all movies.

### **B** Details on the Internet data set for study 1

In France, there are 33 millions local loops, also called connections, that can provide Internet to households, distributed over 15 000 Internet exchange points. Even if all exchange points can provide DSL technology, not all connections qualify for broadband Internet. The download speed of an Internet connection depends on three factors. Firstly, the length of an Internet line reduces its download speed, and homes located far away from an exchange point have lower download speed. Secondly, the diameter of the Internet cable also matters, where larger cables allow for faster connections. Last, and to a lesser extent, the technology embedded into an exchange point can affect the download speed.

A given town can be equipped with one Internet exchange point or with several exchange points if the town is big. Conversely, an Internet exchange point can equip different small towns. An Internet exchange point provides Internet access to users through Internet connections with different speed levels. The coverage rate of high speed Internet infrastructure is defined as the fraction of connections with download speeds higher than 512 kbit/s over the total number of connections delivered by an exchange point. For instance, an Internet exchange point can reach 30,000 households, with 27,000 connections with download speeds exceeding 512 kbit/s, and the remaining connections providing download speed inferior to 512 kbits/s. The coverage rate of this exchange point is 90%. For a given town, the average coverage rate is computed using the rates of all exchange points providing the connections to its inhabitants, weighted by the number of connections supplied by each exchange point. As an example, take a town provided by two exchange points. If the first exchange point can connect 30 000 households and has a high speed Internet coverage rate of 90% while the second one can connect 10 000 households and has a coverage rate of 80%, then the coverage rate of the town is 87.5%.

High speed Internet use rate is the fraction of households subscribed to connections exceeding 512 kbit/s over the total number of connections for an exchange point. The number of households signed up with broadband internet is collected for each exchange point on a quarterly basis by Arcep from Internet service providers. It is linearly interpolated to be used on a monthly basis. Continuing with the example above, with 20 000 households subscribed to broadband internet, the high speed internet use rate is 66.7%. The Internet use rate is aggretated for each town the same way as the Internet coverage rate.

#### Robustness checks for study 2 $\mathbf{C}$

	(1)	(2)	(3)	(4)
	ln_entree1	ln_entree1	ln_entree_seance1	ln_entree_seance1
Hadopi $\times U.S.$	$0.157^{***}$		0.153**	
	(0.0475)		(0.0494)	
Hadopi1 $\times U.S.$		$0.232^{**}$		0.317***
		(0.0788)		(0.0865)
Hadopi2 $\times U.S.$		$0.303^{***}$		$0.335^{***}$
		(0.0644)		(0.0657)
Hadopi $3 \times U.S$ .		0.0993 +		0.102 +
		(0.0572)		(0.0588)
log(Screens)	У	у	n	n
log(Ad)	У	У	У	У
log(Budget)	У	У	У	У
Consumer rating	У	У	У	У
Press rating	У	У	У	У
Genre fe	У	У	У	У
Nationality fe	У	У	У	У
Art and House fe	У	У	У	У
Age restriction fe	У	У	У	У
Month fe	У	У	У	У
Distributor fe	У	У	У	У
N	615	615	615	615
$R^2$	0.915	0.919	0.703	0.713

Table 16: DiD French vs. American in France (first week): demand reaction, an additional robustness test

Robust standard errors + p < 0.10, \* p < 0.05, \*\* p < 0.01, \*\*\* p < 0.001

	(1) log(Admissions)	(2) log(Admissions)	(3) log(Admissions/Screens)	(4) log(Admissions/Screens)
α10				
Hadopi $\times U.S.$	-0.0308 (0.120)		0.00919 (0.117)	
Hadopi 1 $\times U.S.$	()	0.185 (0.207)		0.0649 (0.192)
Hadopi 2 $\times U.S.$		0.101 (0.171)		0.0787 (0.178)
Hadopi 2 $\times U.S.$		0.0367 (0.136)		-0.00280 (0.145)
q25				
Hadopi $\times U.S.$	0.0987 (0.106)		-0.0489 (0.102)	
Hadopi1 $\times U.S.$		0.205 (0.177)	~ /	0.198 (0.171)
Hadopi 2 $\times U.S.$		0.284+ (0.155)		0.175 (0.154)
Hadopi 3 $\times U.S.$		0.0616 (0.119)		-0.0526 (0.124)
q50				
Hadopi $\times U.S.$	0.113 (0.105)		0.120 (0.0968)	
Hadopi1 $\times U.S.$		0.284 (0.180)		0.308 + (0.183)
Hadopi $2 \times U.S$ .		$0.265^{*}$ (0.135)		$0.331^{*}$ (0.142)
Hadopi 3 $\times U.S.$		0.0682 (0.115)		0.0402 (0.118)
q75				
Hadopi $\times U.S.$	$0.221^{*}$ (0.104)		$0.244^{*}$ (0.103)	
Hadopi1 $\times U.S.$		0.196 (0.206)		0.267 (0.216)
Hadopi $2 \times U.S$ .		$0.348^{**}$ (0.125)		$0.341^{*}$ (0.145)
Hadopi $3 \times U.S$ .		0.138 (0.119)		0.138 (0.128)
q90				
Hadopi $\times U.S.$	0.203+ (0.111)		$0.228^{*}$ (0.115)	
Hadopi 1 $\times U.S.$	· · ·	0.253 (0.222)	× ,	$0.496^{*}$ (0.248)
Hadopi 2 $\times U.S.$		$0.279^{*}$ (0.138)		0.350*
Hadopi 3 $\times U.S.$		0.212		0.177
$\frac{N}{R^2}$	767	767	817	767

Table 17: DiD French vs. American in France (first week) : baseline results with quantile treatment effect

Bootstrapped standard errors in parentheses + p < 0.10, \* p < 0.05, \*\* p < 0.01, \*\*\* p < 0.001All regressions control for number of screens, marketing expediture, production budget, consumer and press rating, and include genre fe, nationality fe, art and house fe, age restriction fe, month fe, and distributor fe.

### D Control group selection for study 3

During the period 2008-2011, no measure similar to the French Hadopi law has been implemented in Europe, besides Sweden in 2009. However, three waves of legislation in countries close to France may have stirred up the Internet users community, but we believe they did not have any significant impact on individuals because they have not been implemented. In Spain, in 2010, discussions were surrounding the Sinde Law, a provision designed to shut down websites in violation with copyright law. It was made law at the end of 2011. In Italy, in 2011, the regulation and competition authority for the communication industries, the AGCOM, expressed its intention to fight websites in violation with copyright law. The Spanish and the Italian measures are of quite a different nature compared to the Hadopi law. Finally in the United Kingdom, in 2010, the Digital Economy Act addressed online copyright infringement in a similar way to the graduated response of the Hadopi law. With this act, Internet subscribers who had downloaded illegal content using peer-to-peer file-sharing systems were supposed to receive notifications. This Act is in force although the procedure against infringers has not been implemented.

We then run the following estimation using observations in the pre-treatment period (i.e form January 2007 to March 2009) to select the relevant candidate to be included in the control group:

$$Outcome_{Fr,t} = \alpha + \beta_s Outcome_{ct} + \epsilon_{ct} \tag{7}$$

c is a country and t is a month, with  $Outcome_{ct}$  being repetitively  $Log(Admissions_{ct}/Screens_{ct})$  for US movies,  $Log(Screens_{ct})$  for US movies,  $Lag_{ct}$  for US movies,  $Log(Admissions_{ct})$  for all movies, and  $Log(USmarketshare_{ct})$ .

Table 10.	109(11a)	1110010110[	2/20/00/	<i>oci)</i> ( <i>o</i> anc	arj <b>2</b> 001 - 1		500)	
	(1) Belgium	(2) Finland	(3) Germany	(4) Italy	(5) Netherlands	(6) Spain	(7) Switzerland	(8) UK
Log(Admissions/Screens)	$0.459^{***}$ (0.128)	$0.381^{***}$ (0.116)	0.368*** (0.0962)	0.223 (0.144)	$0.395^{***}$ (0.0795)	$0.432^{*}$ (0.245)	$0.409^{***}$ (0.141)	$0.301^{***}$ (0.0714)
Constant	(0.855)	(0.724)	4.397*** (0.619)	(0.939)	(0.522)	$3.908^{**}$ (1.605)	$4.170^{***}$ (0.897)	(0.497)
R-squared	0.329	0.464	0.396	0.109	0.372	0.301	0.365	0.437

Table 18:  $Log(Admissions_{ct}/Screens_{ct})$  (January 2007 - March 2009)

Standard errors with Newey-West corrections allowing for observations within a country to be correlated up to 6 months. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1.

Our preferred control group when the outcome of interest is the number of admissions over the number of screens of US movies is constituted by Belgium, Finland, Germany, the Netherlands, Spain, Switzerland, and United Kingdom.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Belgium	Finland	Germany	Italy	Netherlands	Spain	Switzerland	UK
Log(Screens)	$0.304^{**}$	0.123 (0.120)	0.139 (0.144)	$-0.168^{*}$	$0.379^{*}$	-0.0672	0.0485 (0.134)	0.0674 (0.193)
Constant	(0.100)	(0.120)	(0.111)	(0.0013)	(0.101)	(0.100)	(0.101)	(5.100)
	$6.262^{***}$	7.464***	$6.946^{***}$	$9.396^{***}$	$5.696^{***}$	8.611***	$7.802^{***}$	$7.512^{***}$
	(0.661)	(0.610)	(1.168)	(0.756)	(1.202)	(1.043)	(0.742)	(1.608)
Observations R-squared	$\begin{array}{c} 27 \\ 0.132 \end{array}$	$\begin{array}{c} 27 \\ 0.020 \end{array}$	$\begin{array}{c} 27 \\ 0.024 \end{array}$	$\begin{array}{c} 27 \\ 0.041 \end{array}$	$\begin{array}{c} 27 \\ 0.090 \end{array}$	$\begin{array}{c} 27 \\ 0.003 \end{array}$	$\begin{array}{c} 27 \\ 0.004 \end{array}$	$\begin{array}{c} 27 \\ 0.003 \end{array}$

(0 ) (1  $\sim$ 

 $Standard\ errors\ with\ Newey-West\ corrections\ allowing\ for\ observations\ within\ a\ country\ to\ be\ correlated\ up\ to\ 6\ months.\ ^{***}p\ <\ 0.01,\ ^{**}p\ <\ 0.05,$ \* p < 0.1.

Our preferred control group when the outcome of interest is the number of screens of US movies is constituted by Belgium, and the Netherlands.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Belgium	Finland	Germany	Italy	Netherlands	Spain	Switzerland	UK
Lag	$0.473^{***}$	0.456	-0.119	$0.322^{***}$	-0.0420	-0.404***	-0.0671	$0.405^{**}$
Constant	(0.118)	(0.271)	(0.166)	(0.104)	(0.0395)	(0.0609)	(0.109)	(0.147)
	$2.565^{**}$	$3.722^{**}$	$8.031^{***}$	$3.282^{**}$	$7.322^{***}$	$11.42^{***}$	$7.443^{***}$	$4.886^{***}$
	(1.241)	(1.694)	(1.960)	(1.468)	(0.804)	(0.820)	(0.761)	(1.058)
Observations R-squared	$\begin{array}{c} 27 \\ 0.172 \end{array}$	$\begin{array}{c} 27 \\ 0.080 \end{array}$	$\begin{array}{c} 27 \\ 0.011 \end{array}$	$27 \\ 0.141$	$\begin{array}{c} 27 \\ 0.004 \end{array}$	$\begin{array}{c} 27 \\ 0.207 \end{array}$	$\begin{array}{c} 27 \\ 0.004 \end{array}$	$27 \\ 0.113$

Table 20:	$Lag_{ct}$ (	January	2007 -	March	2009)
-----------	--------------	---------	--------	-------	-------

Standard errors with Newey-West corrections allowing for observations within a country to be correlated up to 6 months. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1.

Our preferred control group when the outcome of interest is the time lag of US movies is constituted by Belgium, Italy, and United Kingdom.

Our preferred control group when the outcome of interest is the number of admissions of all movies is constituted by Belgium, Germany, the Netherlands, Spain, Switzerland, and United Kingdom.

Our preferred control group when the outcome of interest is the number of admissions over the number of screens of all movies is constituted by Belgium, Germany, Italy, Spain, Switzerland, and United Kingdom.

	raoie i	<b>       _</b>	ramitteeter	(000)	aarj <b>2</b> 001	maron ac	,00)	
	(1) Belgium	(2) Finland	(3) Germany	(4) Italy	(5) Netherlands	(6)Spain	(7) Switzerland	(8) UK
Log(Admissions)	$0.751^{***}$ (0.105)	0.0478 (0.0759)	$0.519^{***}$ (0.0809)	0.0541 (0.0449)	$0.405^{***}$ (0.132)	$0.351^{*}$ (0.174)	$0.398^{***}$ (0.116)	$0.321^{*}$ (0.176)
Constant	$6.056^{***}$ (1.453)	$15.87^{***}$ (0.960)	$8.284^{***}$ (1.306)	$15.64^{***}$ (0.666)	$10.81^{***}$ (1.885)	$10.99^{***}$ (2.738)	$11.10^{***}$ (1.577)	$ \begin{array}{r}     11.34^{***} \\     (2.835) \end{array} $
Observations R-squared	$\begin{array}{c} 27 \\ 0.660 \end{array}$	$\begin{array}{c} 27 \\ 0.003 \end{array}$	$\begin{array}{c} 27 \\ 0.346 \end{array}$	$\begin{array}{c} 27 \\ 0.016 \end{array}$	$27 \\ 0.158$	$\begin{array}{c} 27 \\ 0.073 \end{array}$	$\begin{array}{c} 27 \\ 0.250 \end{array}$	$\begin{array}{c} 27 \\ 0.095 \end{array}$

Table 21: Log(Admissions<sub>ct</sub>) (January 2007 - March 2009)

 $\overline{S}$ tandard errors with Newey-West corrections allowing for observations within a country to be correlated up to 6 months. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1.

Table 22:  $Log(Admissions_{ct}/Screens_{ct})$  (January 2007 - March 2009)

	(1) Belgium	(2) Finland	(3) Germany	(4) Italy	(5) Netherlands	(6) Spain	(7) Switzerland	(8) UK
Log(Admissions/Screens)	1.918***	0.347	1.581***	0.955***	0.389	0.819*	3.370***	2.134**
Constant	(0.410) -0.124*	(0.320) -0.471***	(0.290) - $0.150^{***}$	(0.330) - $0.234^{**}$	(0.288) - $0.485^{***}$	(0.468) - $0.449^{***}$	(0.401) 0.0485	(1.002) - $0.406^{***}$
	(0.0680)	(0.153)	(0.0535)	(0.101)	(0.134)	(0.122)	(0.0583)	(0.136)
Observations R-squared	$\begin{array}{c} 27 \\ 0.684 \end{array}$	$\begin{array}{c} 27 \\ 0.072 \end{array}$	$\begin{array}{c} 27 \\ 0.535 \end{array}$	$27 \\ 0.298$	$\begin{array}{c} 27 \\ 0.041 \end{array}$	$\begin{array}{c} 27 \\ 0.062 \end{array}$	$\begin{array}{c} 27 \\ 0.614 \end{array}$	$\begin{array}{c} 27 \\ 0.109 \end{array}$

Standard errors with Newey-West corrections allowing for observations within a country to be correlated up to 6 months. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1.



Figure 7: Countries included in the control group for the Log(Admissions/Screens) of US movies.

Figure 8: Countries not included in the control group for the Log(Admissions/Screens) of US movies.



Figure 9: Countries included in the control group for the Log(Screens) of US movies.





Figure 10: Countries not included in the control group for the Log(Screens) of US movies.

Figure 11: Countries included in the control group for the time Lag of US movies.





Figure 12: Countries not included in the control group for the time Lag of US movies.

Figure 13: Countries included in the control group for the aggregate number of admissions of all movies.





Figure 14: Countries not included in the control group for the aggregate number of admissions of all movies.

Figure 15: Countries included in the control group for the US market share.





Figure 16: Countries not included in the control group for the US market share.

# **E** Interpretation of $\beta_a$ coefficients for study 4

A simplified version of equation (6) we estimate in Section 5 is the following one :

$$\begin{split} Y_{ijta} = &\gamma_i + \lambda_t + \delta_a + \alpha After_t * USA_i + \rho_1 After * Young1_{ta} + \rho_2 After * Young2_{ta} + \eta_1 USA * Young1_{ta} \\ &+ \eta_2 USA * Young2_{ta} + \beta_1 After * USA * Young1_{ita} + \beta_2 After * USA * Young2_{ita} + \epsilon_{ijta} \end{split}$$

Then, we define six differences:

(1) 
$$E[Y_{ijta}|i = USA, a = Young1, t = After] - E[Y_{ijta}|i = USA, a = Young1, t = Before]$$
  
=  $\gamma_{USA} + \lambda_{After} + \delta_{Young1} + \alpha + \rho_1 + \eta_1 + \beta_1 - \gamma_{USA} - \lambda_{Before} - \delta_{Young1} - \eta_1$   
=  $\lambda_{After} - \lambda_{Before} + \alpha + \rho_1 + \beta_1$ 

(2) 
$$E[Y_{ijta}|i = Fr, a = Young1, t = After] - E[Y_{ijta}|i = Fr, a = Young1, t = Before]$$
  
=  $\gamma_{Fr} + \lambda_{After} + \delta_{Young1} + \rho_1 - \gamma_{Fr} - \lambda_{Before} - \delta_{Young1}$   
=  $\lambda_{After} - \lambda_{Before} + \rho_1$ 

(3) 
$$E[Y_{ijta}|i = USA, a = Young2, t = After] - E[Y_{ijta}|i = USA, a = Young2, t = Before]$$
  
=  $\gamma_{USA} + \lambda_{After} + \delta_{Young2} + \alpha + \rho_2 + \eta_2 + \beta_2 - \gamma_{USA} - \lambda_{Before} - \delta_{Young2} - \eta_2$   
=  $\lambda_{After} - \lambda_{Before} + \alpha + \rho_2 + \beta_2$ 

(4) 
$$E[Y_{ijta}|i = Fr, a = Young2, t = After] - E[Y_{ijta}|i = Fr, a = Young2, t = Before]$$
  
=  $\gamma_{Fr} + \lambda_{After} + \delta_{Young2} + \rho_2 - \gamma_{Fr} - \lambda_{Before} - \delta_{Young2}$   
=  $\lambda_{After} - \lambda_{Before} + \rho_2$ 

(5) 
$$E[Y_{ijta}|i = USA, a = Old, t = After] - E[Y_{ijta}|i = USA, a = Old, t = Before]$$
  
=  $\gamma_{USA} + \lambda_{After} + \delta_{Old} + \alpha - \gamma_{USA} - \lambda_{Before} - \delta_{Old}$   
=  $\lambda_{After} - \lambda_{Before} + \alpha$ 

(6) 
$$E[Y_{ijta}|i = Fr, a = Old, t = After] - E[Y_{ijta}|i = Fr, a = Old, t = Before]$$
  
=  $\gamma_{Fr} + \lambda_{After} + \delta_{Old} - \gamma_{Fr} - \lambda_{Before} - \delta_{Old}$   
=  $\lambda_{After} - \lambda_{Before}$ 

With those six differences, we can give the interpretation of  $\beta_1$ ,  $\beta_2$ , and  $\beta_1 - \beta_2$  using difference-in-difference-in-differences:

$$\beta_1 = [(1) - (2)] - [(5) - (6)] \tag{8}$$

$$\beta_2 = [(3) - (4)] - [(5) - (6)] \tag{9}$$

$$\beta_1 - \beta_2 = [(1) - (2)] - [(3) - (4)] \tag{10}$$

As a result,  $\beta_1$  (resp.  $\beta_2$ )can be interpreted as the difference in Y between consumers aged 10-29 (resp. 30-49) and consumers aged 50-80, between US movies and French movies, after and before the law is implemented.  $\beta_1 - \beta_2$  can be interpreted as the same triple differences, but between people aged 10-29 and people aged 30-49.

### F Theoretical framework

This section presents a stylized model of the consumer choice which provides several predictions that can be tested empirically. We over-simplify the user's choice to emphasize the mechanism involved in the effect of an anti-piracy law as there are already many general approaches in the theoretical literature about file sharing that are cited in the literature review. We note  $w_i$  (resp.  $u_j$ ) the net utility of watching movie illegally online (resp. movie j legally in theaters)

We consider a market wherein individuals can choose to watch up to two movies, i and j. We assume that only a movie i can be illegally watched online, and that consumers face a budget/time constraint which prevent of fraction of consumers from watching both movies in theaters. Let us first consider the movie i. An option for a consumer is to buy a ticket to legally watch a movie i in a theater. Then she will get a utility  $u_i$  comprising the value of the movie, the price of the movie, and the time involved to watch it in a movie theater. Another option is to illegally download the movie on the Internet and watch it at home. In that case, the consumer gets a utility  $w_i$  that comprises the value of the movie and the transaction cost from illegally downloading the movie. The transaction cost from illegally downloading a movie comprises: the time to find and download an acceptable copy, the risk to catch a virus, the chance to be detected as an infringer and to be sentenced, the worse experience of watching a movie at home than in a movie theater (a movie theater allows the user from spending time with friends and benefiting from improved viewing equipments), and the extra value of the illegal consumption (like the absence of travel costs and waiting time, the possibility to stop watching a movie at any time and come back to it later on, and higher ease to stop watching a movie that one does not enjoy). A consumer always has the outside option not to watch a movie i with the utility  $u_0$  that we normalize to zero. The consumer will simply choose the option that maximizes its utility. We note  $L_i$  the legal consumption of movie i,  $D_i$  the illegal consumption of movie i after downloading, and  $\varnothing$  the behavior of consuming no movie. For example, the action  $L_i$ corresponds to the case where a consumer legally buys movie i. It happens if  $u_i \ge w_i$  and if  $u_i \ge 0$ . The utility of downloading decreases once the anti-piracy law is introduced. The consumer can now obtain  $w'_i$ with  $w'_i \leq w_i$ .<sup>29</sup> The same reasoning applies for a movie j: we note  $L_j$  the legal consumption of movie j, and  $\emptyset$  the behavior of consuming no movie (the movie *j* cannot be illegally downloaded).

There are six different behaviors before the law that can lead to eleven different possibilities when an anti-piracy law is enforced. The law has no effect on consumers who already legally consume both movies i and j, on individuals who either only legally watch a movie i or a movie j, and on persons who do not watch any movie. It can only affect users that illegally download. Some consumers illegally watch a movie i and legally purchase a movie j without the law. With the law, they can switch to the legal consumption of both movies if they are not constrained by time or by money. The main contribution of this model is that they may also stop from watching a movie j and switch to the legal consumption of a movie i if they are constrained and if they prefer i to j. Either, they may stop from watching a movie i and keep purchasing a movie j if they are constrained and if they prefer a movie j to a movie i, or if  $u_i < 0$ . We summarize this mechanism in table 23.

<sup>&</sup>lt;sup>29</sup>Economic studies on criminal behavior indicate that individual's reaction to increased enforcement depends on the risk profile of the individual. Risk-seeking individuals may increase their illegal behavior when law enforcement increases (Ehrlich (1973), Heineke (1978)). Fortunately, most individuals are rather risk-adverse.

Defore the law	THE THE	Effect of the legal market
$L_i, L_j$	$L_i, L_j$	No effect
$L_i, \varnothing$	$L_i, \varnothing$	No effect
$\emptyset, L_j$	$\emptyset, L_j$	No effet
	$L_i, L_j$	Market expansion effect (no revenue and no time constraints)
$D \cdot L \cdot$	$L_i, \varnothing$	Business stealing effect (revenue and time contraints)
$D_i, D_j$	$\emptyset, L_j$	No effect
	$D_i, L_j$	No effect
	$L_i, \varnothing$	Market expansion effect
D. Ø	$D_i, \varnothing$	No effect
$D_{i}, \mathcal{D}$	Ø, Ø	No effect
Ø.Ø	ø.ø	No effect

Table 23: Consumer behavior before and after the anti-piracy law Before the law After the law Effect on the legal market

 $L_i$  (resp.  $L_j$ ) corresponds to the legal consumption of movie *i* (resp. *j*),  $D_i$  to the illegal consumption of movie *i* after downloading, and  $\emptyset$  to the behavior of consuming no movie.

The law has no effect on individuals who already legally watch movies i and j, on consumers who only legally watch one of the two movies (either the movie i or the movie j), and on consumers who do not watch any movie, legally or illegally. Users download the movie i and purchase the movie j without the law if  $w_i > u_i$  and  $u_j > 0.^{30}$  With the law, they are converted into legal consumers of both movies if  $u_i > w'_i$ , if  $u_j > 0$ , and if they are not budget/time constrained. If they are budget constrained, then they switch to the legal consumption of the movie i and they stop watching the movie j if  $u_i > w'_i + u_j$  (when  $w'_i < 0$ ) or if  $u_i > u_j$  (when  $w'_i < 0$ ). Remark that whatever the sign of  $w'_i$ ,  $u_i > u_j$  is a necessary condition for this situation to occur. They switch to the legal consumption of the movie j and stop watching the movie i if  $u_j > 0$ ,  $u_i < 0$ , and  $w'_i < 0$  when they are unconstrained, or if  $u_j > 0$ ,  $u_j > u_i$ , and  $w'_i < 0$  when they are constrained. They keep downloading the movie i and buying the movie j if they are unaffected by the law (that is if  $w'_i > u_i$  and  $u_j > 0$  or if  $w'_i + u_j > u_i$ ). Lastly, for users that download the movie i without purchasing the movie j before the law, we have  $w_i > u_i$ ,  $w_i > 0$ , and  $u_j > 0$ , they keep downloading the movie i if  $u_i > w'_i$  and  $u_i > 0$ , and they stop from watching the movie i if  $w'_i < 0$  and  $u_i < 0$ .

Individuals are heterogeneous in the three parameters that define them,  $u_i$ ,  $u_j$ , and  $w_i$ . So the different cases of table 23 correspond to consumer groups with different preferences. Hence, it is an empirical question to determine which effect prevails to test the models. The size of the market expansion effect depends on the proportion of individuals who become legal consumers, i.e., the share of individuals who are persuaded enough by the law to stop downloading movies and who enjoy movies enough to buy them in theater. Similarly, the size of the business stealing effect depends on the proportion of individuals who are convinced by the law to stop downloading movies, who enjoy movies enough to legally buy, that

<sup>&</sup>lt;sup>30</sup>Remark that this situation can happen even if  $u_i > u_j$  and  $u_i > w_i$ , that is even if the consumer strictly prefer legally watching the movie *i* over watching the movie *j* and she strictly prefer legally watching the movie *i* over downloading the movie *i*.

are budget and time constrained, and that prefer the movie i to the movie j. In Section 3, we motivate why a movie i represents an U.S. movie and a movie j a French movie.

Using this model, it is possible to compute an approximation of the loss in consumer welfare in the market for movies in theaters. Without the anti-piracy law, the consumer's utility is given by  $w_i + u_j$ . With the law, it becomes  $u_i$ . The variation in utility induced by the law is simply given by  $w_i + u_j - u_i$ , with  $w_i + u_j > u_i$  by construction. The utility of legal consumption primarily comprises the value of the movie minus the price. The utility of illegal consumption is mainly composed of the value of the movie minus the transaction cost of downloading it illegally. The difference in utility  $u_j - u_i$  cancels out the price term because movies exhibit uniform prices (see Einav and Orbach (2007) for a discussion). The difference  $w_i - u_i$  cancels out the value of the movie *i*. Then, assuming that the transaction cost is close to zero before the law,  $w_i + u_j - u_i$  can be interpreted as the value of a movie j.<sup>31</sup> A lower bound of the value of a movie *j* is given by its price because its value is necessarily superior to its price. Finally, an estimation of the change in welfare is simply given by the variation in admissions to French films times the average ticket price. This loss in consumer welfare is estimated at between 97 and 126 million euros over the period between September 2010 and December 2011, that is, between 73 and 94 million euros per year in the short-term.

This approximation of the change in welfare is quite restrictive for several reasons. We can only estimate the change in welfare for consumers involved in the business stealing effect. We cannot estimate the welfare variation for users who stopper downloading movies but who were not converted into legal consumers because of a lack of data. Moreover, our estimation of welfare loss says nothing about how consumers fill their new free time. In particular, we do not consider subsequent substitution with DVDs or Blu-rays, as we have no information on the video market. Finally, this welfare loss is a short-term estimation. If, as aimed by the law, fighting against piracy fosters in the long term the production of high quality movies, a dynamic analysis of the welfare could have a different result. For instance, Giorcelli and Moser (2015) show that the adoption of copyright laws within Italy has increased the number and quality of new operas.

<sup>&</sup>lt;sup>31</sup>The assumption that the transaction is close to zero seems plausible for the following reasons: the risk to be detected as an infringer before the law was close to zero, and, for a fraction of consumers, the extra value of watching illegally a movie at home might compensate the extra value of watching a movie in a well-equipped film theater.

Liste des documents de travail de la Direction des Études et Synthèses Économiques

G 9001	J. FAYOLLE et M. FLEURBAEY Accommission, profitabilité et endettement des		Macro-economic import functions with imperfect competition - An application to the E.C. Trade
G 9002	enreprises H. ROUSSE Détection et effets de la multicolinéarité dans les modèles linéaires ordinaires - Un prolongement de la réflexion de BELSLEY, KUH et WELSCH	G 9203	<ol> <li>STAPIC</li> <li>STAPIC</li> <li>Starbard and the services de la France dans le cadre des négociations multila- térales du GATT</li> <li>Juin 1992 (1ère version)</li> </ol>
G 9003	P. RALLE et J. TOUJAS-BERNATE Indexation des salaires : la rupture de 1983	G 9204	Novembre 1992 (version finale) P. SEVESTRE
G 9004	D. GUELLEC et P. RALLE Compétitivité, croissance et innovation de produit		L'économétrie sur données individuelles- temporelles. Une note introductive
G 9005	P. RALLE et J. TOUJAS-BERNATE Les conséquences de la désindexation. Analyse dans une maquette prix-salaires	G 9205	H. ERKEL-ROUSSE Le commerce exterieur et l'environnement in- ternational dans le modèle AMADEUS (réésstimation 1992)
G 9101	Équipe AMADEUS Le modèle AMADEUS - Première partie - Présentation générale	G 9206	N. GREENAN et D. GUELLEC Coordination within the firm and endogenous growth
G 9102	J.L. BRILLET Le modèle AMADEUS - Deuxième partie - Propriétés variantielles	G 9207	A. MAGNIER et J. TOUJAS-BERNATE Technology and trade: empirical evidences for the major five industrialized countries
G 9103	D. GUELLEC et P. RALLE Endogenous growth and product innovation	G 9208	B. CREPON, E. DUGUET, D. ENCAOUA et P. MOHNEN
G 9104	<ul> <li>H. ROUSSE</li> <li>Le modèle AMADEUS - Troisième partie - Le commerce extérieur et l'environnement international</li> </ul>	G 9209	Cooperative, non cooperative R & D and optimal patent life B. CREPON et E. DUGUET Research and develooment, competition and
G 9105	<ul> <li>H. ROUSSE</li> <li>Effets de demande et d'offre dans les résultats du commerce extérieur manufacturé de la France au</li> </ul>		innovation: an application of pseudo maximum likelihood methods to Poisson models with heterogeneity
G 9106	cours des deux dernieres decennies B. CREPON Innovation, taille et concentration : causalités et dynamiques	G 9301	<ol> <li>TOUJAS-BERNATE</li> <li>Commerce international et concurrence impar- faite : développements récents et implications pour la politique commerciale</li> </ol>
G 9107	<ul> <li>B. AMABLE et D. GUELLEC</li> <li>Un panorama des théories de la croissance endogène</li> </ul>	G 9302	Ch. CASES Durées de chômage et comportements d'offre de travail : une revue de la littérature
G 9108	M. GLAUDE et M. MOUTARDIER Une évaluation du coût direct de l'enfant de 1979 à 1989	G 9303	<ul> <li>H. ERKEL-ROUSSE</li> <li>Union économique et monétaire : le débat économique</li> </ul>
G 9109	P. RALLE et alii France - Allemagne : performances économiques comparées	G 9304	N. GREENAN - D. GUELLEC / G. BROUSSAUDIER - L. MIOTTI Innovation organisationnelle, dynamisme tech- nologique et performances des entrieprises
G 9110	J.L. BRILLET Micro-DMS NON PARU	G 9305	P. JAILLARD P. JAILLARD Le traité de Maastricht : précentation initidiciue et
G 9111	<ul> <li>A. MAGNIER</li> <li>Effets accélérateur et multiplicateur en France depuis 1970 : quelques résultats empiriques</li> </ul>	G 9306	historique J.L. BRILLET Micro DMS - ordenation of monthlyde
G 9112	B. CREPON et G. DUREAU Investissement en recherche-développement : analyse de causalités dans un modèle d'accélé-	G 9307	Micro-DMS - Presentation of proprietos J.L. BRILLET Micro-DMS - variantes : les tableaux
G 9113	rateur generalise J.L. BRILLET, H. ERKEL-ROUSSE, J. TOUJAS- BERNATE	G 9308	<ul> <li>S. JACOBZONE</li> <li>Les grands réseaux publics français dans une perspective européenne</li> </ul>
	"France-Allemagne Couplées" - Deux économies vues par une maquette macro-économétrique	G 9309	L. BLOCH - B. CŒURE Profitabilité de l'investissement productif et
G 9201	W.J. ADAMS, B. CREPON, D. ENCAOUA Choix technologiques et stratégies de dissuasion d'entrée	G 9310	transmission des chocs financiers J. BOURDIEU - B. COLIN-SEDILLOT Los théorice sur lo erructure coningle du conital -
G 9202	J. OLIVEIRA-MARTINS, J. TOUJAS-BERNATE		Les ineories sur la succure oprintate du capital . quelques points de repère

G 9311	<ol> <li>BOURDIEU - B. COLIN-SEDILLOT Les décisions de financement des entreprises françaises : une évaluation empirique des théo- ries de la structure optimale du capital</li> </ol>	G 9412	analyse économique des politiques allemande J. BOURDIEU - B. CŒURÉ - B. COI IN-SFDI II OT
G 9312	L. BLOCH - B. CŒURÉ Q de Tobin marginal et transmission des chocs financiers		Investissement, incertitude et irrév. Quelques développements récents de l'investissement
G 9313	Équipes Amadeus (INSEE), Banque de France, Métric (DP) Présentation des propriétés des principaux mo-	G 9413	B. DORMONT - M. PAUCHET L'évaluation de l'élasticité emploi-s elle des structures de qualification
G 9314	B. CREPON - E. DUGUET B. Screench & Development, competition and innovation	G 9414 G 9501	I. KABLA Le Choix de breveter une invention J. BOURDIEU - B. CŒURÉ - B. SI
G 9315	<ul> <li>B. DORMONT Quelle est l'influence du coût du travail sur l'emploi ?</li> </ul>	G 9502	When is there a Value of Waiting? L. BLOCH - B. CŒURÉ Imperferions du merché du crédit
G 9316	D. BLANCHET - C. BROUSSE Deux études sur l'âge de la retraite	C 05/13	minperieduors du marche du credu, ment des entreprises et cycle écon D. GOLIY - E. MALIENN
G 9317	<ul> <li>D. BLANCHET</li> <li>Répartition du travail dans une population hété- rogène : deux notes</li> </ul>	0000	U. GOOA - E. MAJUNIN Les transformations de la demand qualification en France Une étude sur la période 1970-199
G 9318	D. EYSSARTIER - N. PONTY AMADEUS - an annual macro-economic model for the medium and long term	G 9504	N. GREENAN Technologie, changement organis: lifications et emploi : une étude em
G 9319	G. CETTE - Ph. CUNÉO - D. EYSSARTIER - J. GAUTIÉ Les effets sur l'emploi d'un abaissement du coût du travail des jeunes	G 9505	l'industrie manufacturiere D. GOUX - E. MAURIN Persistance des hiérarchies sector laires: un réexamen sur données fr
G 9401	D. BLANCHET Les structures par âge importent-elles ?	G 9505 Bis	D. GOUX - E. MAURIN Persistence of inter-industry wages
G 9402	<ol> <li>GAUTIÉ</li> <li>Le chômage des jeunes en France : problème de Icomation ou phénomène de file d'attente ? Quelques éléments du débat</li> </ol>	G 9506	S. JACOBZONE S. JACOBZONE Les liens entre RMI et chômage, u perspective
G 9403	P. QUIRION Les déchets en France : éléments statistiques et économiques	G 9507	NON PARU - article sorti dans Ecc Prévision n° 122 (1996) - pages 95 G. CETTE - S. MAHFOUZ
G 9404	D. LADIRAY - M. GRUN-REHOMME Lissage par moyennes mobiles - Le problème des extrémités de série	G 9601	Le partage primaire du revenu Constat descriptif sur longue pério Banque de France - CEPREMAP -
G 9405	V. MAILLARD Théorie et pratique de la correction des effets de jours ouvrables		Prévision - Érasme - INSEE - OFC Structures et propriétés de cinq mc économiques français
G 9406	F. ROSENWALD La décision d'investir	G 9602 G 9603	Rapport d'activité de la DESE de l' J. BOURDIEU - A. DRAZNIEKS
G 9407	<ul> <li>S. JACOBZONE</li> <li>Les apports de l'économie industrielle pour définir la stratègie économique de l'hôbital public</li> </ul>	G 9604	L'octroi de crédit aux PME : une ar d'informations bancaires A. TOPIOL-BENSAÏD
G 9408	L. BLOCH, J. BOURDIEU, B. COLIN-SEDILLOT, G. LONGUEVILLE Du défaut de palement au dépôt de bilan : les banquiers face aux PME en difficulté	G 9605	Les implantations japonaises en Fr P. GENIER - S. JACOBZONE Comportements de prévention, cor d'alcool et tabagie : peut-on parler
G 9409	D. EYSSARTIER, P. MAIRE Impacts macro-économiques de mesures d'aide	9090 0	globale du capital santé ? Une modélisation microéconométr
G 9410	au logeriren - querques elerrients o evaluador F. ROSENWALD Suivi conjoncturel de l'investissement	0000	C. UCZ - F. LENGLAN Factor analysis and unobserved co models: an application to the study business surveys
G 9411	C. DEFEUILLEY - Ph. QUIRION Les déchets d'emballages ménagers : une	G 9607	N. GREENAN - D. GUELLEC La théorie coopérative de la firme

	analyse économique des politiques française et
412	<ul> <li>allentatuce</li> <li>J. BOURDIEU - B. CŒURÉ -</li> <li>B. COLIN-SEDILLOT</li> <li>Restissement, incertitude et irréversibilité Quelques développements récents de la théorie de l'investissement</li> </ul>
413	B. DORMONT - M. PAUCHET L'évaluation de l'élasticité emploi-salaire dépend- elle des structures de qualification ?
414	I. KABLA Le Choix de breveter une invention
501	<ol> <li>BOURDIEU - B. CŒURÊ - B. SEDILLOT Irreversible Investment and Uncertainty: When is there a Value of Waiting?</li> </ol>
502	L. BLOCH - B. CŒURÉ Imperfections du marché du crédit, investisse- ment des entreprises et cycle économique
503	D. GOUX - E. MAURIN Les transformations de la demande de travail par qualification en France Une étude sur la période 1970-1993
504	<ul> <li>N. GREENAN</li> <li>Technologie, changement organisationnel, qua- lifications et emploi : une étude empirique sur l'industrie manufacturière</li> </ul>
505	D. GOUX - E. MAURIN Persistance des hiérarchies sectorielles de sa- laires: un réexamen sur données françaises
505 is	D. GOUX - E. MAURIN Persistence of inter-industry wages differentials: a reexamination on matched worker-firm panel data
206	S. JACOBZONE Les liens entre RMI et chômage, une mise en perspective perspective article sorti dans Économie et Pevision n° 122 (1996) - pages 85 à 113
507	<ul> <li>G. CETTE - S. MAHFOUZ</li> <li>Le partage primaire du revenu</li> <li>Constat descriptif sur longue période</li> </ul>
601	Banque de France - CEPREMAP - Direction de la Prévision - Érasme - INSEE - OFCE Structures et propriétés de cinq modèles macro- économiques français
602	Rapport d'activité de la DESE de l'année 1995
603	<ol> <li>BOURDIEU - A. DRAZNIEKS L'octroi de crédit aux PME : une analyse à partir d'informations bancaires</li> </ol>
604	A. TOPIOL-BENSAÏD Les implantations japonaises en France
605	P. GENIER - S. JACOBZONE Comportements de prévention, consommation d'alcoof et abagie : peut-on parler d'une gestion globale du capital santé ? Une modélisation microéconométrique empirique
606	C. DOZ - F. LENGLART Factor analysis and unobserved component models: an application to the study of French

9608	N GREENAN - D GUIELLEC	6 9714	
	Technological innovation and employment reallocation	5	Does the Fr state Inflatio
G 9609	Ph. COUR - F. RUPPRECHT L'intégration asymétrique au sein du continent américain : un essai de modélisation	G 9715	X. BONNET Peut-on me baisse des
G 9610	S. DUCHENE - G. FORGEOT - A. JACQUOT Analyse des évolutions récentes de la producti- vité apparente du travail	G 9716	Une etude s N. IUNG - F Productivité
G 9611	X. BONNET - S. MAHFOUZ The influence of different specifications of wages- prices spirals on the measure of the NAIRU: the case of France	G 9717	ď échelle da français E. DUGUE1 Appropriatic
G 9612	PH. COUR - E. DUBOIS, S. MAHFOUZ, J. PISANI-FERRY The osof fristed retrenchment revisited: how strond is the evidence?	G 9718	the patent s analysis at t L.P. PELÉ - Ârre de la re
G 9613	<ul> <li>A. JACQUOT</li> <li>Les flexions des taux d'activité sont-elles seule- ment conjoncturelles ?</li> </ul>	G 9719	général ZHANG Yin Lexique ma
G 9614	ZHANG Yingxiang - SONG Xueqing Lexique macroéconomique Français-Chinois	G 9720	chinois-fran M. HOUDEI
G 9701	<ul> <li>J.L. SCHNEIDER</li> <li>La taxe professionnelle : éléments de cadrage économique</li> </ul>	G 9721	A. MOURO
G 9702	<ul> <li>J.L. SCHNEIDER Transition et stabilité politique d'un système redistributif</li> </ul>	G 9722	Credibilite, Une revue ( P. AUGER/
G 9703	D. GOUX - E. MAURIN Train or Pay: Does it Reduce Inequalities to En- courage Firms to Train their Workers?		Les donnée Le système Passage de sectorielles
G 9704	P. GENIER Deux contributions sur dépendance et équité	G 9723	P. AUGER/ Using Busir
G 9705	E. DUGUET - N. IUNG R & D Investment, Patent Life and Patent Value An Econometric Analysis at the Firm Level	G 9724	Accounts: tl P. AUGER/ Les compte
G 9706	M. HOUDEBINE - A. TOPIOL-BENSAÏD Les entreprises internationales en France : une analyse à partir de données individuelles		sage aux co d'entreprise <i>paraître</i>
G 9707	<ul> <li>M. HOUDEBINE</li> <li>Polarisation des activités et spécialisation des départements en France</li> </ul>	G 9801	H. MICHAL Présentatio
G 9708	<ul> <li>E. DUGUET - N. GREENAN</li> <li>Le biais technologique : une analyse sur données individuelles</li> </ul>		Une étude ( pour la Frar
G 9709	J.L. BRILLET Analyzing a small French ECM Model	G 9803	A. BUNNE Apports et l « Real Busi
G 9710	<ul> <li>J.L. BRILLET Formalizing the transition process: scenarios for capital accumulation</li> </ul>	G 9804	C. BARLET D. ENCAOI The Comm
G 9711	<ul> <li>G. FORGEOT - J. GAUTIÉ Insertion professionnelle des jeunes et processus de déclassement</li> </ul>	G 9805	French mar P. CAHUC
G 9712	E. DUBOIS High Real Interest Rates: the Consequence of a Saving Investment Disequilibrium or of an in- sufficient Credibility of Monetary Authorities?	G 9806	Power - Evi Equalizing / Power - Evi J. ACCARD
G 9713	Bilan des activités de la Direction des Études et Synthèses Économiques - 1996		La producti 1996
	_		

Ξ

	F. LEQUILLER Does the French Consumer Price Index Over- state Inflation?	G 9807	Bilan des activités de la Direction d Synthèses Économiques - 1997
	X. BONNET X. BONNET Peut-on mettre en évidence les rigidités à la	G 9808	A. MOUROUGANE Can a Conservative Governor Con comodative Monetary Policy?
	baisse des salaires nominaux ? Une étude sur quelques grands pays de l'OCDE N ILING - F_RUPPRECHT	G 9809	X. BONNET - E. DUBOIS - L. FAU Asymétrie des inflations relatives e · tests sur l'inflation francaise
	Productivité de la recherche et rendements d'échelle dans le secteur pharmaceutique français	G 9810	E. DUGUET - N. IUNG Sales and Advertising with Spillove level: Estimation of a Dynamic Stru-
	E. DUGUEL 1 - I. KABLA Appropriation strategy and the motivations to use the patient system in France - An econometric analysis at the firm level	G 9811	on Panel Data J.P. BERTHIER Congestion urbaine : un modèle de pointe à courbe débit-vitesse et dei
	L.P. PELÉ - P. RALLE Âge de la retraite : les aspects incitatifs du régime général	G 9812	élastique C. PRIGENT La nart des salaires dans la valeur
_	ZHANG Yingxiang - SONG Xueqing Lexique macroéconomique français-chinois, chinois-français	G 9813	approche macroéconomique A.Th. AERTS
_	M. HOUDEBINE - J.L. SCHNEIDER Mesurer l'influence de la fiscalité sur la locali- sation des entreprises	0001	L evolution de la part des saraires o ajoutée en France reflète-t-elle les individuelles sur la période 1979-19 Β ε ελι λλιιτέ
	<ul> <li>A. MOUROUGANE</li> <li>Créchbilité, indépendance et politique monétaire</li> <li>Une revue de la littérature</li> </ul>	G 9901	S. DUCHÊNE - A. JACQUOT S. DUCHÊNE - A. JACQUOT
	P. AUGERAUD - L. BRIOT Les données comptables d'entreprises Le système intermédiaire d'entreprises Passage des données individuelles aux données	G 9902	One croissance prus nome en imp début de tá décennie ? Une analys raison internationale Ch. COLIN Modélisation des carrières dans De
	P. AUGERAUD - J.E. CHAPRON P. AUGERAUD - J.E. CHAPRON Using Business Accounts for Compiling National Accounts: the French Experience	G 9903	Ch. COLIN Évolution de la dispersion des sala de prospective par microsimulation
	P. AUGERAUD Les comptes d'entreprise par activités - Le pas-	G 9904	B. CREPON - N. IUNG Innovation, emploi et performances
	sage aux comptes - De la comptabilité d'entreprise à la comptabilité nationale - A paraître	G 9905	B. CREPON - Ch. GIANELLA Wages inequalities in France 1969 An application of quantile regressic
	H. MICHAUDON - C. PRIGENT Présentation du modèle AMADEUS J. ACCARDO Une étude de comptabilité dénérationnelle	G 9906	C. BONNET - R. MAHIEU Microsimulation techniques applied generational transfers - Pensions ir framework: the case of France
	pour la France en 1996 X. BONNET - S. DUCHÊNE Apports et limites de la modélisation	G 9907	F. ROSENWALD L'impact des contraintes financière cision d'investissement
	« Real Business Cycles »	G 9908	Bilan des activités de la DESE - 19
	C. BARLET - C. DUGUET - D. ENCAOUA - J. PRADEL The Commercial Success of Innovations An econometric analysis at the firm level in	G 9909	J.P. ZOYEM Contrat d'insertion et sortie du RMI Évaluation des effets d'une politiqu
	French manufacturing P. CAHUC - Ch. GIANELLA -	G 9910	Ch. COLIN - FI. LEGROS - R. MAH Bilans contributifs comparés des ré retraite du secteur privé et de la for
	о. GOUX - А. ZILDERBERG Equalizing Wage Differences and Bargaining Power - Evidence form a Panel of French Firms	G 9911	G. LAROQUE - B. SALANIÉ Une décomposition du non-emploi
	<ul> <li>J. ACCARDO - M. JLASSI La productivité globale des facteurs entre 1975 et 1996</li> </ul>	G 9912	B. SALANIÉ Une maquette analytique de long te marché du travail
		G 9912	Ch. GIANELLA

,		
/ités de la Direction des Études et onomiques - 1997	Bis	Une estimation de l'élasticité de l'emploi peu qualifié à son coût
5ANE vative Governor Conduct an Ac- onetary Policy?	G 9913	Division « Redistribution et Politiques Sociales » Le modèle de microsimulation dynamique DESTINIE
E. DUBOIS - L. FAUVET inflations relatives et menus costs ation française	G 9914	E. DUGUET Macro-commandes SAS pour l'économétrie des panels et des variables qualitatives
N. IUNG reftising with Spillovers at the firm on of a Dynamic Structural Model	G 9915	R. DUHAUTOIS Évolution des flux d'emplois en France entre 1990 et 1996 : une étude empirique à partir du fichier des bénéfices réels normaux (BRN)
R baine : un modèle de trafic de e débit-vitesse et demande	G 9916	<ul> <li>J.Y. FOURNIER</li> <li>Extraction du cycle des affaires : la méthode de Baxter et King</li> </ul>
laires dans la valeur ajoutée : une	G 9917	B. CRÉPON - R. DESPLATZ - J. MAIRESSE Estimating price cost margins, scale economies and workers' bargaining power at the firm level
roéconomique la part des salaires dans la valeur ince reflète-t-elle les évolutions	G 9918	Ch. GIANELLA - Ph. LAGARDE Productivity of hours in the aggregate production function: an evaluation on a panel of French firms from the manufacturing sector
ur la période 1979-1994 ? e des séries non-stationnaires	G 9919	<ol> <li>S. AUDRIC - P. GIVORD - C. PROST Évolution de l'emploi et des coûts par quali- fication entre 1982 et 1996</li> </ol>
<ul> <li>- A. JACQUOT</li> <li>e plus riche en emplois depuis le cennie ? Une analyse en compa-</li> </ul>	G 2000/01	R. MAHIEU Les déterminants des dépenses de santé : une approche macroéconomique
ionale les carrières dans Destinie	G 2000/02	C. ALLARD-PRIGENT - H. GUILMEAU - A. QUINET The real exchange rate as the relative price of nontrables in terms of tradables: theoretical investigation and eminimal study on Fench data
a dispersion des salaires : un essai » par microsimulation N. IUNG	G 2000/03	<ul> <li>JY. FOURNIER L'approximation du filtre passe-bande proposée par Christiano et Fitzgerald</li> </ul>
plot et periotinances Ch. GIANFLLA	G 2000/04	Bilan des activités de la DESE - 1999
itties in France 1969-1992 of quantile regression techniques R. MAHIEU	G 2000/05	B. CREPON - F. ROSENWALD Investissement et contraintes de financement : le poids du cycle Une estimation sur données françaises
ansfers - Pensions in a dynamic e case of France	G 2000/06	<ul> <li>A. FLIPO</li> <li>Les comportements matrimoniaux de fait</li> </ul>
LD ontraintes financières dans la dé- issement	G 2000/07	R. MAHIEU - B. SÉDILLOT Microsimulations of the retirement decision: a supply side approach
rités de la DESE - 1998	G 2000/08	<ul> <li>C. AUDENIS - C. PROST</li> <li>Déficit conjoncturel : une prise en compte des conjonctures passées</li> </ul>
rtion et sonte du Kwi s effets d'une politique sociale 1. LEGROS - R. MAHIEU	G 2000/09	R. MAHIEU - B. SÉDILLOT Équivalent patrimonial de la rente et souscription de retraite complémentaire
anis compares de la fonction publique - B. SALANIÉ stirion du non-emploi en France	G 2000/10	R. DUHAUTOIS Ralentissement de l'investissement : petites ou grandes entreprises ? industrie ou tertiaire ?
analytique de long terme du vail	G 2000/11	G. LAROQUE - B. SALANIÉ Temps partiel féminin et incitations financières à l'emploi
4	G2000/12	Ch. GIANELLA Local unemployment and wages

	C. AUDENIS - P. BISCOURP - N. RIEDINGER Existe-t-il une asymétrie dans la transmission du prix du brut aux prix des carburants ? F. MAGNIEN - JL. TAVERNIER - D. THESMAR	Les statistiques internationales de PIB par habitant en standard de pouvoir d'achat: une analyse des feutats Bina das achivités de In DESE - 2000	Diani des acuivies de la DESE - 2001 B. SÉDILLOT - E. WALRAET	La cessation d'activité au sein des couples : y a-t- il interdépendance des choix ? G. BRILHAULT	<ul> <li>Rétropolation des séries de FBCF et calcul du capital fixe en SEC-95 dans les comptes nationaux francais</li> </ul>	<ul> <li>Retropolation of the investment series (GFCF) and estimation of fixed capital stocks on the ESA-95 basis for the French balance sheets</li> <li>P. BISCOURP - B. CRÉPON - T. HECKEL - N.</li> </ul>	NIEUNGEK How do firms respond to cheaper computers? Microsconometric evidence for France based on a production function approach	L'impact des nouvelles technologies de l'information et de la communication sur	l'economie française - un bouclage macro- économique	<ol> <li>J. BARDAJI - B. SÉDILLOT - E. WALRAET Évaluation de trois rétormes du Régime Général d'assurance vieillesse à l'aide du modèle de</li> </ol>	microsimulation DESTINIE JP. BERTHIER Réflexions sur les différentes notions de volume	dans les comptes nationaux: comptes aux prix d'une année fixe ou aux prix de l'année récédente, séries chaînées	P. HILU F. Rilcu réponses des entreprises aux enquêtes de conjoncture ?	<ol> <li>ROBERT-BOBÉE</li> <li>Romportements démographiques dans le Les comportements démographiques dans le modèle de microsimulation Destinie - Une comparaison des estimations issues des enquiets Jeunes et Carrières 1997 et Histoire</li> </ol>	rammaue 1999 JP. ZOYEM La dynamique des bas revenus : une analyse des entrées-sorties de pauvreté	F. HILD Prévisions d'inflation pour la France M. LECLAIR Réduction du temps de travail et tensions sur les	E. WALRAET - A. VINCENT E. WALRAET - A. VINCENT - Analyse de la redistribution intragénérationnelle dans le système de retraite des salariés du privé - line anoroche nar minorient	- Intragenerational distributional analysis in the Intragenerational distributional analysis in the french private sector pension scheme - A microsimulation approach
~	G2001/17 G2002/01	G2002/02	G2002/03 G2002/03	G2002/04		G2002/05		00/20020		G2002/07	G2002/08		62002/09	G2002/10	G2002/11	G2002/12 G2002/13	G2002/14	
	<ul> <li>B. CREPON - Th. HECKEL</li> <li>Informatisation en France : une évaluation à partir de données individuelles</li> <li>Computerizitation in France: an evaluation based on individual common data</li> </ul>	<ul> <li>La nouvelle économie et la mesure de la croissance du PIB</li> </ul>	<ul> <li>The new economy and the measure ment of GDP growth</li> </ul>	S. AUDRIC La reprise de la croissance de l'emploi profite-t- elle aussi aux non-diplômés ?	<ol> <li>BRAUN-LEMAIRE Évolution et répartition du surplus de productivité</li> </ol>	A. BEAUDU - Th. HECKEL Le canal du crédit fonctionne-t-il en Europe ? Une étude de l'hétérogéněité des comportements dinvestissement à partir de données de bilan adrédees	C. AUDENIS - P. BISCOURP - N. FOURCADE - O. LOISEL Testing the augmented Solow growth model: An empirical reassessment using panel data	R. MAHIEU - B. SÉDILLOT Départ à la retraite, irréversibilité et incertitude	Bilan des activités de la DESE - 2000	<ul> <li>J. Ph. GAUDEMET</li> <li>Les dispositifs d'acquisition à titre facultatif d'annutés viagères de retraite</li> </ul>	B. CRÉPON - Ch. GIANELLA Fiscalité, coût d'usage du capital et demande de facteurs : une analyse sur données individuelles	B. CRÉPON - R. DESPLATZ Évaluation des effets des dispositifs d'allegements de channes sorciales sur les has salaries	JY. FOURNIER Comparaison des salaires des secteurs public et privé	<ul> <li>JP. BERTHIER - C. JAULENT</li> <li>R. CONVENEVOLE - S. PISANI</li> <li>R. CONSTRUCTOR</li> <li>R. eméthodologie de comparaison entre Une méthodologie de comparaison intermédiaires de source fiscale et de comptabilité nationale</li> </ul>	P. BISCOURP - Ch. GIANELLA Substitution and complementarity between capital, skilled and less skilled workers: an analysis at the firm level in the French	Interrutacioning indusity I. ROBERT-BOBEE Modelling demographic behaviours in the French microsimulation model Destinie: An analysis of microsimulation model dentifity	JP.ZOYEM Diagnostic sur la pauvreté et calendrier de revenus : le cas du "Panel européen des ménages »	JY. FOURNIER - P. GIVORD La réduction des taux d'activité aux âges extrêmes, une spécificité française ?
	G2000/13	G2001/01		G2001/02	G2001/03	G2001/04	G2001/05	G2001/06	G2001/07	G2001/08	G2001/09	G2001/10	G2001/11	G2001/12	G2001/13	G2001/14	G2001/15	G2001/16

P. CHONE - D. LE BLANC - I. ROBERT-BOBEE Offre de travail féminine et garde des jeunes enfants	F. MAUREL - S. GREGOIR Les indices de compétitivité des pays : inter- prétation et limites	N. RIEDINGER - E.HAUVY Le coût de dépollution atmosphérique pour les entreprises françaises : Une estimation à partir de données individuelles	P. BISCOURP et F. KRAMARZ Création d'emplois, destruction d'emplois et internationalisation des entreprises industrielles françaises : une analyse sur la période 1986- 1992	Bilan des activités de la DESE - 2002	PO. BEFFY - J. DEROYON - N. FOURCADE - S. GREGOIR - N. LAÏB - B. MONERT Evolutions dérmographiques et croissance : une projection macro-économique à l'horizon 2020	P. AUBERT La situation des salariés de plus de cinquante ans dans le secteur privé	P. AUBERT - B. CRÉPON Age, salaire et productivité La productivité des salariés décline-t-elle en fin de carrière ?	H. BARON - P.O. BEFFY - N. FOURCADE - R. MAHIEU Le ralentissement de la productivité du travail au cours des années 1990	PO. BEFFY - B. MONFORT Patrimoine des ménages, dynamique d'allocation et comportement de consommation	P. BISCOURP - N. FOURCADE Peur-on mettre en évidence l'existence de rigidités à la baisse des sataires à partir de réonnées individuelles ? Le cas de la France à la fin des années 90	M. LECLAIR - P. PETIT Présence syndicale dans les firmes : quel impact sur les inégalités salariales entre les hommes et les femmes ?	PO. BEFFY - X. BONNET - M. DARRACQ- PARIES - B. MONFORT MZE: a small macro-model for the euro area	P. AUBERT - M. LECLAIR La compétitivité exprimée dans les enquêtes trimestrielles sur la situation et les perspectives dans l'industrie	M. DUÉE - C. REBILLARD La dépendance des personnes âgées: une projection à long terme	S. RASPILLER - N. RIEDINGER Régulation environnementale et choix de localisation des groupes français	A. NABOULET - S. RASPILLER Les déterminants de la décision d'investir : une approche par les perceptions subjectives des firmes
G2002/15	G2002/16	G2003/01	G2003/02	G2003/03	G2003/04	G2003/05	G2003/06	G2003/07	G2003/08	G2003/09	G2003/10	G2003/11	G2004/01	G2004/02	G2004/03	G2004/04

G2006/10 G2006/11	G2006/12	G2006/13	G2006/14	G2006/15	G2007/01 G2007/02	G2007/03	G2007/04	G2007/05	G2007/06	G2007/07	G2007/08	G2007/09	G2007/10	6200//11	G2007/12	62008/01
N. FERRARI Prévoir l'investissement des entreprises Un indicateur des révisions dans l'enquête de conjoncture sur les investissements dans l'industrie.	PO. BEFFY - C. L'ANGEVIN Chômage et boucle prix-salaires : apport d'un modèle « qualifiés/peu qualifiés »	B. HEITZ A two-states Markov-switching model of inflation in France and the USA: credible target VS inflation spiral	O. BIAU - H. ERKEL-ROUSSE - N. FERRARI Réponses individuelles aux enquêtes de conjoncture et prévision macroéconomiques : Exemple de la prévision de la production Exemple de la prévision de la production	P. AUBERT - D. BLANCHET - D. BLAU The labour market after age 50: some elements of a Franco-American comparison	D. BLANCHET - T. DEBRAND - P. DOURGNON - P. POLLET L'enquête SHARE : présentation et premiers résultats de l'édition française	M. DUÉE La modélisation des comportements démogra- phiques dans le modèle de microsimulation DESTINIE	H. RAOUI - S. ROUX Étude de simulation sur la participation versée aux salariés par les entreprises	C. BONNET - S. BUFFETEAU - P. GODEFROY Disparités de retraite de droit direct entre hommes et femmes : quelles évolutions ?	C. PICART Les gazelles en France	P. AUBERT - B. CRÉPON - P. ZAMORA Le rendement apparent de la formation continue dans les entreprises : effets sur la productivité et les salaires	JF. OUVRARD - R. RATHELOT Demographic change and unemployment: what do macroeconometric models predict?	D. BLANCHET - JF. OUVRARD Indicateurs d'engagements implicites des systemes de retraite : chriftrages, propriétés analytiques et réactions à des chocs admographiques types	G. BIAU - O. BIAU - L. ROUVIERE Nonparametric Forecasting of the Manufacturing Output Growth with Firm-level Survey Data	C. AFSA - P. GIVORD Le rôle des conditions de travail dans les absences pour maladie	P. SILLARD - C. L'ANGEVIN - S. SERRAVALLE Performances comparées à l'exportation de la France et de ses principaux partenaires Une analyse structurelle sur 12 ans	X. BOUTIN - S. QUANTIN Une méthodologie d'évaluation comptable du coût du capital des entreprises françaises : 1984- 2002
G2005/09	G2005/10	G2005/11	G2005/12	G2005/13	G2005/14	G2005/15	G2005/16	G2006/01	G2006/02	G2006/03	G2006/04	G2006/05	G2006/06	G2006/07	G2006/08	G2006/09

1 C. LELARGE Les entrenrises (industrielles) francaises sont-	G2008/03	The Effects of Retail Regulations on Prices Evidence form the Loi Galland Y. BARBSDL - A. BRIANT Economiss of anothim fartion of modulativité des
Les entreprises (industrielles) françaises sont- elles à la frontière technologique ? O. BIAU - N. FERRARI Théorie de l'opinion Faut-il pondérer les réponses individuelles ?	G2008/04	Economies d'agglomération et productivité des entreprises : estimation sur données individuelles françaises D. BLANCHET - F. LE GALLO Les projections démographiques : principaux
<ul> <li>A. KOUBI - S. ROUX</li> <li>D. ne reinterprétation de la relation entre productivité et inégalités salariales dans les entreprises</li> </ul>	G2008/05	mecanismes et recour sur le experience irançaise DE LANCHET - F.: TOUTLEMONDE Evolutions démographiques et déformation du cycle de vie active : quelles relations ?
R. RATHELOT - P. SILLARD The impact of local taxes on plants location decision	G2008/06	M. BARLET - D. BLANCHET - L. CRUSSON Internationalisation et flux d'emplois : que dit une approche comptable ?
L. GONZALEZ - C. PICART Diversification, recentrage et poids des activités de support dans les groupes (1993-2000)	G2008/07	C. LELARGE - D. SRAER - D. THESMAR Entrepreneurship and Credit Constraints - Evidence from a French Loan Guarantee Program
D. SKAEK Allègements de cotisations patronales et dynamique salariale	G2008/08	X. BOUTIN - L. JANIN Are Prices Really Affected by Mergers?
V. ALBOUY - L. LEQUIEN Les rendements non monétaires de l'éducation : le cas de la santé	G2008/09	M. BARLET - A. BRIANT - L. CRUSSON Concentration géographique dans l'industrie manufacturitie et dans les services en France: une approche par un indicateur en continu
D. BLANCHE I - I. DEBKAND Aspiration à la retraite, santé et satisfaction au travail : une comparaison européenne	G2008/10	M. BEFY - É. COUDIN - R. RATHELOT Who is confronted to insecure labor market histories? Some evidence based on the French
M. BARLET - L. CRUSSON Quel impact des variations du prix du pétrole sur la croissance française ?	G2008/11	labor market transition M. ROGER - E. WALRAET Social Security and Well-Being of the Elderly: the
C. PICART Flux d'emploi et de main-d'œuvre en France : un réexamen	G2008/12	Case of France C. AFSA
V. ALBOUY - C. TAVAN Massification et démocratisation de l'enseignement supérieur en France		Analyser les composantes du bien-être et de son volution Une approche empirique sur données individuelles
T. LE BARBANCHON The Changing response to oil price shocks in France: a DSGE type approach	G2008/13	M. BARLET - D. BLANCHET - T. LE BARBANCHON Microsimuler le marché du travail : un prototype
T. CHANEY - D. SRAER - D. THESMAR Collateral Value and Corporate Investment Evidence from the French Real Estate Market	G2009/01	PA. PIONNIER Le partage de la valeur ajoutée en France, 1949-2007
J. BOISSINOT Consumption over the Life Cycle: Facts for France	G2009/02	Laurent CLAVEL - Christelle MINODIER A Monthly Indicator of the French Business Climate
C. AFSA Interpréter les variables de satisfaction : l'exemple de la durée du travail	G2009/03	H. ERKEL-ROUSSE - C. MINODIER Do Business Tendency Surveys in Industry and Services Help in Forecasting GDP Growth?
R. RATHELOT - P. SILLARD Zones Franches Urbaines: quels effets sur l'emploi salarié et les créations d'établissements ?	G2009/04	A Real-Time Analysis on French Data P. GIVORD - L. WILNER Les contrats temporaties : trappe ou marchepied vers l'embloi stable ?
V. ALBOUY - B. CRÉPON Aléa moral en santé: une évaluation dans le cadre du modéle causal de Rubin	G2009/05	G. LALANNE - PA. PIONNIER - O. SIMON Le partage des fruits de la croissance de 1950 à 2008 : une approche par les comptes de surplus
C. PICART Les PWE françaises : rentables mais peu dynamiques	G2009/06	L. DAVEZIES - X. D'HAULTFOEUILLE Faut-il pondérer ? Ou l'éternelle question de l'économètre confronté à des données d'enquête

S. QUANTIN - S. RASPILLER - S. SERRAVALLE Commerce intragroupe, fiscalité et prix de V. BELLAMY - G. CONSALES - M. FESSEAU -S. LE LANDIER - É. RAYNAUD Une décomposition du compte des ménages de la comptabilité nationale par catégorie de Detecting Economic Regimes in France: a Qualitative Markov-Switching Indicator Using Hierarchical Positions - Evidence on French Full-Time Male Workers from a matched Employer-Employee Dataset M. CLERC - V. MARCUS Élasticités-prix des consommations énergétiques D. BLANCHET - J. LE CACHEUX - V. MARCUS Adjusted net savings and other approaches to I. BUONO - G. LALANNE The Effect of the Uruguay round on the Intensive N. CECI-RENAUD - P.-A. CHEVALIER Les seuils de 10, 20 et 50 salariés : impact sur la Méthodes économétriques pour l'évaluation de politiques publiques Prix du pétrole et croissance potentielle à long L'IPC, miroir de l'évolution du coût de la vie en Les trajectoires professionnelles en début de vie Discrimination à l'embauche : comment exploiter valeurs publiées et des séries des valeurs révisées - Un exercice de prévision en temps réel Health Expenditure Models: a Comparison of Five France ? Ce qu'apporte l'analyse des courbes National Origin Differences in Wages and Tome 1 – Version avec volumes à prix constants transferts : une analyse sur données françaises Avantages comparés des séries des premières de la croissance trimestrielle du PIB en France active : quel impact des contrats temporaires ? Le modèle MÉSANGE réestimé en base 2000 FOUGÈRE Partage de la valeur ajoutée, approche données microéconomiques G. LALANNE - E. POULIQUEN - O. SIMON Y. BARBESOL - P. GIVORD - S. QUANTIN sustainability: some theoretical background V. ALBOUY - L. DAVEZIES - T. DEBRAND AEBERHARDT - D. Specifications using Panel Data and Extensive Margins of Trade R. AEBERHARDT - J. POUGET taille des entreprises françaises M.-É. CLERC - É. COUDIN les procédures de testing ? S. BLASCO - P. GIVORD J. BARDAJI - F. TALLET Mixed Frequency Data C. KLEIN - O. SIMON ménage en 2003 **R. RATHELOT** des ménages C. MINODIER P. GIVORD d'Engel terme Ŀ. G2009/10 G2009/12 G2009/13 G2009/14 G2009/15 G2009/07 G2009/09 G2010/02 G2010/04 G2010/05 G2010/06 G2010/07 G2010/08 G2009/08 G2009/11 G2010/03 G2010/01

par

viii

Ξ

	P. CHONÉ - F. EVAIN - L. WILNER - E. YILMAZ Introducing activity-based payment in the hospital industry : Evidence from French data	C. GRISLAIN-LETRÉMY Natural Disasters: Exposure and Underinsurance P-Y. CABANNES - V. COTTET - Y. DUBOIS - C. I FLARRES - M. SUCSIC	French Firms in the Face of the 2008/2009 Crisis A. POISSONNIER - D. ROY Households Satellite Account for France in 2010. Methodological issues on the assessment of	domestic production G. CLÉAUD - M. LEMOINE - PA. PIONNIER Which size and evolution of the government expenditure multiplier in France (1980-2010)?	M. BACHELET - A. LEDUC - A. MARINO Les biographies du modèle Destinie II : rebasage et projection	B. GARBINTI L'achat de la résidence principale et la création d'entreprises sont-ils favorisés par les donations et héritages ?	<ul> <li>N. CECI-RENAUD - P. CHARNOZ - M. GAINI Évolution de la volatilité des revenus salariaux du secteur privé en France depuis 1968</li> <li>P. ALIRERT</li> </ul>	Modalités d'application des réformes des retraites et prévisibilité du montant de pension C. GRISLAIN-LETRÉMY - A. KATOSSKY	The Impact of Hazardous Industrial Facilities on Housing Prices: A Comparison of Parametric and Semiparametric Hedonic Price Models JM. DAUSSIN-BENICHOU - A. MAUROUX	Turning the heat up. How sensitive are households to fiscal incentives on energy efficiency investments?	C. LABONNE - G. LAMÉ Credit Growth and Capital Requirements: Binding or Not? C. GRISLAIN-LETRÉMY et C. TREVIEN	The Impact of Housing Subsidies on the Rental Sector: the French Example M. LEQUIEN et A. MONTAUT	Croissance potentielle en France et en zone euro: un tour d'horizon des méthodes d'estimation B. GARBINTI - P. LAMARCHE Les hauts revenus épargnent-lis davantage ?	D. AUDENAERT - J. BARDAJI - R. LARDEUX - M. ORAND - M. SICSIC Wage Resilience in France since the Great Recession	F. ARNAUD - J. BOUSSARD - A. POISSONNIER - H. SOUAL Computing additive contributions to growth and other issues for chain-linked curaterly accreates	H. FRAISSE - F. KRAMARZ - C. PROST Labor Disputes and Job Flows
×	G2013/11	G2013/12 G2013/13	G2013/14	G2013/15	G2014/01	G2014/02	G2014/03 G2014/04	G2014/05	G2014//06		G2014/07 G2014/08	G2014/09	G2014/10	G2014/11	G2014/12	G2014/13
	A. EIDELMAN - F. LANGUMIER - A. VICARD Prélèvements obligatoires reposant sur les ménogos : des canaux redistributifs différents en	1990 et zOTO O. BARGAIN - A. VICARD Le RMI et son successeur le RSA découragent- ils certains jeunes de travailler ? Une analyse sur	les jeunes autour de 25 ans C. MARBOT - D. ROY Projections du coût de l'APA et des craractéristiques de ses bénéficiaires à l'horizon 2040 à l'aide du modelle Destinie	A. MAUROUX Le crédit d'impôt dédié au développement durable : une évaluation économétrique	V. COTTET - S. QUANTIN - V. REGNIER Coti du travail et allègements de charges : une estimation au niveau établissement de 1996 à 2008	X. D'HAULTFOEUILLE - P. FÉVRIER - L. WILNER Demand Estimation in the Presence of Revenue Management	D. BLANCHET - S. LE MINEZ D. BLANCHET - S. LE MINEZ Joint macro/micro evaluations of accrued-to-date pension liabilities: an application to French reforms	<ol> <li>DEROYON - A. MONTAUT - P-A PIONNIER Utilisation rétrospective de l'enquéte Emploi à une fréquence mensuelle : apport d'une</li> </ol>	modelisation espace-état C. TREVIEN Habiter en HLM : quel avantage monétaire et quel impact sur les conditions de logement ?	<ul> <li>A. POISSONNIER Temporal disaggregation of stock variables - The Chow-Lin method extended to dynamic models</li> </ul>	P. GIVORD - C. MARBOT Does the cost of child care affect female labor market participation? An evaluation of a French reform of childcare subsidies	G. LAME - M. LEQUIEN - PA. PIONNIER Interpretation and limits of sustainability tests in public finance	C. BELLEGO - V. DORTET-BERNADET La participation aux pôles de compétitivité : quelle inidance sur les dépenses de R&D et l'activité des PME et ETI ? P - Y. CARANNES - A. MONTALIT -	PA. PIONNIER Evaluer la productivité globale des facteurs en France : l'apport d'une mesure de la qualité du capital et du travail	R. AEBERHARDT - C. MARBOT Evolution of Instability on the French Labour Market During the Last Thirty Years	-B. BERNARU - G. CLEAUD Oil price: the nature of the shocks and the impact on the French economy G. LAME Was there a « Greenspan Conundrum » in the Euro area?
	G2012/08	G2012/09	G2012/10	G2012/11	G2012/12	G2012/13	G2012/14	G2013/01- F1301	G2013/02- F1302	G2013/03	G2013/04	G2013/05	G2013/06 G2013/07		G2013/08	G2013/09 G2013/10
	M. CLERC - M. GAINI - D. BLANCHET Recommendations of the Stiglitz-Sen-Fitoussi Report: A few illustrations	M. BACHELET - M. BEFFY - D. BLANCHET Projeter l'impact des réformes des retraites sur l'activité des 55 ans et plus : une comparaison de trois modèles	C. LOUVOT-RUNAVOT L'évaluation de l'activité dissimulée des entre- prises sur la base des contrôles fiscaux et son insertion dars les comptes nationaux	A. SCHREIBER - A. VICARU La tertiarisation de l'économie française et le ralentissement de la productivité entre 1978 et 2008	ME. CLERC - O. MONSO - E. POULIQUEN Les inégalités entre générations depuis le baby- boom	C. MARBOT - D. ROY Évaluation de la transformation de la réduction dimpôt en crédit d'impôt pour l'emploi de salariés à domicile en 2007	P. GIVORD - R. RATHELOT - P. SILLARD Place-based tax exemptions and displacement effects: An evaluation of the Zones Franches Urbaines program	X. DHAULTFOEUILLE - P. GIVORD - X. BOUTIN The Environmental Effect of Green Taxation: the Case of the French "BonusMalus"	<ul> <li>M. BARLET - M. CLERC - M. GARNEO - V. LAPÈGUE - V. MARCUS</li> <li>La nouvelle version du modèle MZE, modèle macroéconométrique pour la zone euro</li> </ul>	R. AEBERHARDT - I, BUONO - H. FADINGER Learning, Incomplete Contracts and Export Dynamics: Theory and Evidence form French	Firms C. KERDRAIN - V. LAPÈGUE Restrictive Fiscal Policies in Europe: What are the Likely Effects?	P. GIVORD - S. QUANTIN - C. TREVIEN A Long-Term Evaluation of the First Generation of the French Urban Enterprise Zones	N. CECI-RENAUD - V. COTTET Politique salariale et performance des entreprises P. FÉVRIER - L. WILNER Do Contrauents Corrrectly Expect Price Redurcinors 7 tresting Duvaranic Behavior	M. GAINI - A. LEDUC - A. VICARD School as a sheller? School leaving-age and the business cycle in France	M. GAINI - A. LEDUC - A. VICARD A scarred generation? French evidence on young people entering into a tough labour market P. AUBERT - M. BACHEL FT	Disparities de moritant de pension et redistribution dans le système de retraite français R. AEBERHARDT - P GIVORD - C. MARBOT Spillover Effect of the Minimum Wage in France: An Unconditional Quantile Regression Approach
ý	G2011/07	G2011/08	G2011/09	62011/10	G2011/11	G2011/12	G2011/13	G2011/14	G2011/15	G2011/16	G2011/17	G2012/01	G2012/02 G2012/03	G2012/04	G2012/05 G2012/06	G2012/07
ĸi	PY. CABANNES - V. LAPÈGUE - E. POULIQUEN - M. BEFFY - M. GAINI Quelle croissance de moyen terme après la	crise ? I. BUONO - G. LALANNE La réaction des entreprises françaises à la baisse des tarris douaniers étrangers	R. RATHELOT - P. SILLARD L'apport des méthodes à noyaux pour mesurer la concentration géographique - Application à la concentration des immigrés en France de 1968 à 1999	M. BARATON - M. BEFFY - D. FOUGÈRE Une évaluation de l'effet de la réforme de 2003 sur les départs en retraite - Le cas des enseignants du second denté public.	D. BLANCHET - S. BUFFETEAU - E. CRENNER S. LE MINEZ Le modèle de microsimulation Destinie 2:	principales caractéristiques et premiers résultats D. BLANCHET - E. CRENNER Le bloc retraites du modèle Destinie 2 : duide de l'utilisateur	<ul> <li>M. BARLET - L. CRUSSON - S. DUPUCH -</li> <li>M. BARLET - L. CRUSSON - S. DUPUCH -</li> <li>F. PUECH</li> <li>Des services échangés aux services échan- ceables : une apolication sur données francaises</li> </ul>	M. BEFFY - T. KAMIONKA Public-private wage gaps: is civil-servant human capital sector-specific?	PY. CABANNES - H. ERKEL-ROUSSE - G. LALANNE - O. MONSO - E. POULIQUEN Le modèle Mésange réestimé en base 2000 Tome 2 - Version avec volumes à prix chaînés	R. AEBERHARDT - L. DAVEZIES Conditional Logit with one Binary Covariate: Link between the Static and Dynamic Cases	T. LE BARBANCHON - B. OURLIAC - O. SIMON Les marchés du travail français et américain face aux chocs conjoncturels des années 1986 à 2007 : une modélisation DSGE	C. MARBOT Une évaluation de la réduction d'impôt pour l'emploi de salariés à domicile	L. DAVEZIES Modèles à effets fixes, à effets aléatoires, modèles mixtes ou multi-niveaux: propriétés et mises en œuvre des modélisations de l'hétérogénéité dans le cas de données groupées	M. ROGER - M. WASMER Heterogeneity matters: labour productivity differentiated by age and skills	JL. BKILONGNE - JM. FUUKNIEK V. LAPÈGUE - O. MONSO De la crise financière à la crise économique L'imp la crise perturbations financières de 2007 et 2008 sur la croissance de sept pays industrialisés	P. CHARNOZ - É. COUDIN - M. GAINI Wage inequalities in France 1976-2004: a quantile regression analysis
	G2010/09	G2010/10	G2010/11	G2010/12	G2010/13	G2010/14	G2010/15	G2010/16	G2010/17	G2010/18	G2011/01	G2011/02	G2011/03	G2011/04	620111/05	G2011/06

	J. BARDAJI - JC. BRICONGNE - B. CAMPAGNE - G. GAULIER Compared performancess of French companies on the domestic and foreign markets C. BELI FGO - R. DE NILS.	The redistributive effect of online piracy on the box office performance of American movies in foreign markets															
i	G 2015/16 G 2015/17																
X	P. GIVORD - C. GRISLAIN-LETRÉMY - H. NAEGELE How does tuel taxation impact new car purchases? An evaluation using French consumer-level dataset	P. AUBERT - S. RABATÉ Durée passée en carrière et durée de vie en retraite : quel partage des gains d'espérance de vie ?	A. POISSONNIER The walking dead Euler equation Addressing a challenge to monetary policy models	Y. DUBOIS - A. MARINO Indicateurs de rendement du système de retraite français	T. MAYER - C. TREVIEN The impacts of Urban Public Transportation: Evidence from the Paris Region	S.T. LY - A. RIEGERT Measuring Social Environment Mobility	M. A. BEN HALIMA - V. HYAFIL-SOLELHAC M. KOUBI - C. REGAERT Quel est l'impact du système d'indemnisation maladie sur la durée des arrêts de travail pour maladie ?	Y. DUBOIS - A. MARINO Disparités de rendement du système de retraite dans le secteur privé : approches intergénéra- tionnelle et intragénérationnelle	B. CAMPAGNE - V. ALHENC-GELAS - JB. BERNARD No evidence of financial accelerator in France	Q. LAFFÉTER - M. PAK Élasticités des recettes fiscales au cycle économique : étude de trois impôts sur la période 1979-2013 en France	<ul> <li>JM. DAUSSIN-BENICHOU, S. IDMACHICHE, A. LEDUC et E. POULIQUEN Les déterminants de l'attractivité de la fonction publique de l'État</li> </ul>	P. AUBERT La modulation du montant de pension selon la durée de carrière et l'âge de la retraite : quelles disparités entre assurés ?	V. DORTET-BERNADET - M. SICSIC Effet des aides publiques sur l'emploi en R&D dans les petites entreprises	S. GEORGES-KOT Annual and lifetime incidence of the value-added tax in France	M. POULHÉS Are Enterprise Zones Benefits Capitalized into Commercial Property Values? The French Case	<ul> <li>JB. BERNARD - Q. LAFFÉTER Effet de l'activité et des prix sur le revenu salarial des différentes catégories socioprofessionnelles</li> </ul>	C. GEAY - M. KOUBI - G de LAGASNERIE Projections des dépenses de soins de ville, construction d'un module pour Destinie
	G2014/14	G2014/15	G2015/01	G2015/02	G2015/03	G2015/04	G2015/05	G2015/06	G2015/07	G2015/08	G2015/09	G2015/10	G2015/11	G2015/12	G2015/13	G 2015/14	G 2015/15